"We have a once in a lifetime chance to renew our idea of what a company is for..."

REINVENTING THE FIRM

William Davies



First published in 2009 © Demos. Some rights reserved Magdalen House, 136 Tooley Street, London, SE1 2TU, UK

ISBN 978 1 906693-22-0 Copy edited by Justine Doody, Dublin Series design by modernactivity Typeset by Chat Noir Design, Charente Printed by Lecturis, Eindhoven

Set in Gotham Rounded and Baskerville 10 Cover paper: Flora Gardenia Text paper: Munken Premium White



Mixed Sources Product group from well-managed forests, controlled sources and recycled wood or fiber www.fsc.org Cert no. CU-COC-804101 © 1996 Forest Stewardship Council

REINVENTING THE FIRM

William Davies

DEMOS

Open access. Some rights reserved.

As the publisher of this work, Demos wants to encourage the circulation of our work as widely as possible while retaining the copyright. We therefore have an open access policy which enables anyone to access our content online without charge.

Anyone can download, save, perform or distribute this work in any format, including translation, without written permission. This is subject to the terms of the Demos licence found at the back of this publication. Its main conditions are:

- · Demos and the author(s) are credited
- · This summary and the address www.demos.co.uk are displayed
- \cdot The text is not altered and is used in full
- \cdot The work is not resold
- · A copy of the work or link to its use online is sent to Demos

You are welcome to ask for permission to use this work for purposes other than those covered by the licence. Demos gratefully acknowledges the work of Creative Commons in inspiring our approach to copyright. To find out more go to www.creativecommons.org



Contents

	Acknowledgements	7
	Foreword	9
	Introduction	13
1	The four horsemen of the financial apocalypse	21
2	In praise of pluralism	37
3	A better model of the firm	61
4	Inside the employee-owned firm	77
5	Enabling a new model of the firm	91
	Notes	101
	References	113

Acknowledgements

Thanks first and foremost to the John Lewis Partnership for supporting this work. Particular thanks go to Sam Hinton-Smith, Charlotte Cool, Patrick Lewis and Sally McCombie.

Secondly, I would like to thank all the attendees of the round table discussion held in May 2009 to review the early findings and arguments that were later developed into this report. These were: Phillip Blond, Peter Bradwell, Patrick Burns, Nita Clarke, David Coats, Sam Hinton-Smith, Hannah Jameson, Patrick Lewis, Jonathan Michie and Howard Reed.

Thirdly, thanks go to all those who offered critical feedback and research support over the course of the project. I'd particularly like to thank Patrick Burns of the Employee Ownership Association for his regular assistance with research and introductions, Nigel Mason of Baxi Partnership for his expert assistance, Andrew Pendleton at York University, Jonathan Michie at Oxford University and Howard Reed for their valuable feedback on the draft.

I'm also grateful to those who facilitated the three case studies. Particular thanks to Barry Cooke at Make, Steve Parfett and his co-directors at Parfetts and David Hodgkinson and his team at Quintessa.

Finally, many thanks to all those at Demos who have contributed to and offered support on this project. In particular Peter Bradwell, Peter Harrington, Julia Margo and Richard Reeves deserve thanks for making the project happen. Philip Raymond and Davina Reid offered excellent research support during their Demos internships, for which I am also grateful.

William Davies August 2009

Foreword

This September marks the anniversary of the collapse of Lehman Brothers, when the world's biggest bankruptcy plunged the financial markets into freefall and more than £50 billion was wiped off the FTSE 100 index in a single day.

Thankfully, economic Armageddon has been avoided. As stability has been returning to the financial system, it is human nature to hope that normal service will be resumed and that our economy will return to the benign conditions that prevailed before the credit crunch. Yet many of the economic assumptions of the past 20 years have been decisively swept away.

Capitalism is not in meltdown but it is at a critical juncture, as William Davies argues in this significant contribution to the debate about the future of the firm and its relationship with society. For many years it has been the received wisdom that the pursuit of 'shareholder value' was the best way to motivate management and maximise value for shareholders. The crisis has exposed the weaknesses of the drive for short-term maximum gain.

It is only now that the urgency of addressing the banking system has abated that business leaders, policy makers, commentators and citizens have begun to reflect on what alternative types of capitalist structures might be more inclusive of all stakeholders, be more resilient in the long term and reduce the risk of future crises.

Greater diversity in the way companies are owned and run should be welcomed. As this report highlights, there are multiple ownership models and corporate governance structures that can generate wealth as well as positive benefits for society and shareholders alike.

Employee ownership is one solution to the problem of building a more sustainable economy built on long-term

foreword

foundations. That does not mean blunting the entrepreneurial spirit – far from it. Employee ownership can also help fulfil the increasing desire we have for more influence in our work, reflecting the greater choice we have come to expect in our personal lives, so as to unleash our potential and productivity.

The UK employee-owned sector is worth £25 billion annually and is a growing force in the economy. It has the potential to contribute significantly to the long-term well being of employees and communities as well as to greater social cohesion.

That is why the John Lewis Partnership is proud to have supported this project – and why we welcome Demos' call to government to highlight employee ownership and take the lead in providing incentives and removing barriers to the growth of the sector.

Employee ownership may not be right for every business but it has worked for the John Lewis Partnership for 80 years. I believe it can help us to build an economy in which employees and others have a real stake, and in which success is judged not just by short-term returns but by long-term sustainable performance.

Charlie Mayfield is Chairman of the John Lewis Partnership.

Introduction

The economic crisis that peaked with the banking meltdown of autumn 2008 has been accompanied by some striking renunciations of faith. In October of that year, Alan Greenspan, former Chairman of the US Federal Reserve, declared that 'the whole intellectual edifice' on which financial regulation had rested had collapsed, meaning that he had been 'partially wrong' in his policy decisions.¹ The following March, Jack Welch, the former CEO of General Electric and poster child of the 'shareholder value' movement, admitted that 'shareholder value is the dumbest idea in the world'.²

Greenspan's confession made greater headlines, but Welch's was no less significant. The shareholder value creed – the belief that a company's primary purpose is to maximise its value for the benefit of external shareholders – was as much a part of the neoliberal 'intellectual edifice' as Greenspan's belief that financial markets are self-correcting. The creed, embodied by Welch and articulated in Alfred Rappaport's 1986 book *Creating Shareholder Value*, derived from a very limited understanding of what a firm is. At its heart, the shareholder value philosophy presented firms as comparable to any other economic object that could be owned, traded, invested in and profited from. From this perspective, the price of a firm's stock was the best and most complete representation of its true value, and satisfying shareholder interests was the overriding goal of management.

This perspective blankly ignores a number of critical aspects of how firms work and succeed. It fails to address the fact that the value of a contemporary business consists largely of intangible assets, rooted in people, relationships, intellectual property (IP) and reputation, none of which is easily captured in the quantitative calculations of external investors. Treating these assets like physical items of property – tradable, dispensable,

introduction

swiftly exploitable – can do great damage to their long-term value, as so much evidence on mergers and acquisitions suggests.³ Most significantly, the shareholder value perspective excluded what JK Galbraith referred to as 'power, the great black hole of economics'.⁴ Firms differ in how they distribute power internally, with management, employees, shareholders, customers and other stakeholders all requiring some form of recognition and reward for their on-going commitment to the collective venture. There are *political* choices to be made.

Once this is acknowledged, two things follow. Firstly, that the *range of models* for a firm's ownership and control is far broader than the shareholder value movement ever accepted. This report aims to highlight the diversity of ownership and governance structures that has been largely forgotten about over the last 30 years. This is a celebration of organisational pluralism, and a call for more of it. Politicians, managers and business experts need jolting from the financial and organisational monoculture of recent times.

Secondly, it follows that the politics of the firm *does not take place in a vacuum.* Many of the dominant social, economic and political challenges that Britain will face over the coming years can be exacerbated or alleviated depending on how companies behave and are organised. In addition to producing wealth, firms have a profound influence on well-being and on the fabric of civil society. Different types of organisation produce different levels of inequality and fulfilment in people. It has long been convenient to ignore this, and leave firm structures out of public debate about our society and political economy. The next chapter contains evidence of four areas in which dominant practices and assumptions about the firm are contributing to social, economic and political malaises.

Why now?

The banking crisis represents the most explosive example of how the shareholder value model failed, both as a mechanism for accountability and for value creation. But just as there are those who believe that the fundamentals of the financial system simply need tweaking, there are those who hope that a new improved version of the shareholder value ideal will be sufficient for the post-2008 era. Perhaps greater transparency, shareholder activism and corporate social responsibility are required to offset the risks associated with the arms-length, rapid turnover ownership model. Given that every holder of a private pension or ISA is, indirectly, a 'capitalist', then maybe the system simply needs to be made to work *even harder* on behalf of these ultimate owners.

This report takes a different view. Times of crisis never leave things as they were, but produce a different future out of the wreckage of the past.⁵ Most of the ingredients of the next epoch of economic structures are already present, but will be pieced together anew by a combination of economic, cultural, social and political forces. As far as the ownership and governance of firms is concerned, it would be absurd to suggest that the publicly owned and traded company, with a managerial eye on its share price, is about to go into terminal decline. But at the same time it would be unimaginative and historically naïve to believe that it must necessarily remain the dominant fixture of any capitalist economy.

Capitalism is at a critical juncture, and so is our politics. The fiscal environment is suddenly very different from what it was, and a general election is coming in 2010. Regardless of their exact spending plans, all three parties are conscious of the fact that the creation and distribution of wealth and well-being is not entirely within the state's control. People must be supported, encouraged and empowered to form the social and economic relationships that they desire, such that the state is not society's dominant source of collectivism and security. It is peculiar in this respect that politicians do not have more to say about alternative ownership and organisational structures, which can often exert a significant and positive influence over social outcomes, relieving the strain on the state. New Labour has been prepared to support and celebrate 'social enterprise' as a decentralised force for good in society, which points to the available role for government in nurturing alternative forms of economic organisation, but been strangely silent on the structure of mainstream profit-making bodies.

introduction

This economic and political context creates an exciting opportunity to rethink the firm, its ownership, its management structures and its mechanisms of accountability. What's more, in a situation of some turbulence, we have an opportunity to expand our analytical framework beyond narrow economic concepts of allocative efficiency and performance to include broader social and political considerations. While doing this, we get to address the question of what form of *autonomy* we want in our economy.

Autonomy is a helpfully ambiguous value, straddling economics and politics. There is ample economic evidence concerning the benefits that worker autonomy can deliver to business performance, meaning that there are tangible economic reasons for managers to empower and liberate their employees. A particular idea of consumer autonomy, meanwhile, is at the heart of how markets are designed and regulated. But there is also an emerging *political* framework that seeks to deliver greater autonomy in our economic lives (including in our workplaces) as a component part of a society less liable to domination.⁶

Economic and political accounts of autonomy are not reducible to each other, and can potentially come into conflict. For instance, a firm that sought to empower employees purely as a means to extract greater effort and productivity from them could be accused of excessive work intensification, and would scarcely be satisfying the democratic requirement for nondomination in the workplace. On the other hand, a firm that privileged workplace democracy at the expense of productivity would be failing in an equal and opposite sense. The task, this report argues, is to build firms with a suitable balance between economic and political forms of participation, liberty and empowerment. In the best cases discussed in this report, this balance is struck in such a way as to uphold both sets of priorities.

The shareholder value era had its own answer to this question. Finance should have the maximum freedom and power to dictate how companies were run. Employees should be governed by managers and managers dictated to by shareholders. In fact this model was never honestly realised, as the next chapter will demonstrate. Behind the rhetoric of shareholder value was the autonomy of senior executives to inflate their remuneration with stock options and to pursue short-term routes to profit maximisation.

In the coming years, Britain must experiment with new models of the firm, resting on different varieties of economic autonomy. The most sustainable, engaging and productive forms of future economic organisation will only emerge thanks to a plurality of organisational forms and business models. These will produce new accommodations between our society's respective economic, social and political priorities. As Geoff Mulgan notes of the current crisis:

in some sectors the slump will give new momentum to the old idea that workers should employ capital rather than vice versa.... In other sectors, too, there has been a long-term trend towards more people wanting work to be an end as well as a means, a source of fulfilment as well as earnings.⁷

Possibilities are now broader and the isolation of economics from sociology and politics is less plausible.

This report has a particular focus on employee ownership as a model of the firm, specifically because this appears to offer an optimal model of autonomy, combining high levels of economic performance with politically progressive forms of accountability and governance. The virtue of employee ownership straddles economic and political rationales. As we will see, there is no single template or creed underpinning employee ownership, but instead various financial, managerial and cultural ingredients that are recombined in a variety of ways; there is no employee ownership 'movement'. 'Public interest companies', 'social enterprises', cooperatives and mutuals are additional important parts of Britain's industrial and post-industrial makeup. They offer diverse means of producing, serving, employing, cooperating and owning. What's more, with sufficient imagination and expertise they can be combined to form new, hybrid models.

introduction

Outline of the report

The report is structured as follows. Chapter 1 discusses four dimensions of the UK's critical situation, each of which strengthens the case for exploring new models of ownership and corporate governance. Firstly, there is the banking crisis, which highlights the problems of accountability associated with dominant ownership structures. Secondly, there is a competitiveness crisis, inasmuch as Britain cannot continue to rely on financial services and the housing market as its primary sources of prosperity. This poses questions about the organisations and policies that will support long-term wealth creation in the future. Thirdly, there is the fiscal crisis that imperils progressive ends unless new, non-state-oriented routes to social and economic security can be found. Fourthly, there is a moral crisis, associated with spiralling inequality and a sense of an illegitimate 'winner-take-all' ethic at work in contemporary capitalism, which is then compensated for by an unsustainable culture of debt-fuelled consumerism. Company structures and practices are not innocent in this regard.

Chapter 2 explores ownership and governance of firms in greater detail. It challenges the narrowly economic vision of ownership as the right to control and profit from an asset. Instead it looks at the firm as a political entity, whereby various intangible assets – people, knowledge, relationships, reputation – are organised into some sort of power structure. On closer inspection, to claim that anybody completely owns such assets makes little sense. Alternative political-economic means of organising them are worth considering, and the chapter presents evidence on the plurality of ownership and governance forms in the UK today, highlighting their respective merits. It also points towards emerging foundations and techniques of cooperation that look set to grow over the coming decades.

Chapter 3 presents an examination of employee ownership. It looks at the forms of employee ownership that are possible, including 'direct' share ownership, worker cooperatives and employee benefit trusts. These have both democratic and economic merits, as a basis for empowerment and for greater productivity. There is good reason to believe that the two can reinforce one another, in a virtuous circle of political-economic autonomy.

Chapter 4 delves further into employee ownership, with case studies of three employee-owned firms. These have all been either founded or converted in the last ten years. The firms have been selected to highlight the diversity of the sector and the multiple reasons that exist for employee ownership, and potential benefits that stem from it. The implications are drawn out, and located within the crises described in chapter 1.

Chapter 5 outlines an agenda to diversify Britain's corporate governance and ownership practices. This is a curious policy challenge, since much of the innovation that occurs in this territory cannot be anticipated or prescribed by government. Accountants, lawyers and business schools all have a role to play in broadening assumptions about the optimal ownership and governance structures of organisations. Businesses and owners must also rethink the firm. But government can lead in highlighting the range of options and using the tax system to both highlight and incentivise them.

1 The four horsemen of the financial apocalypse

Consider a scenario. The world power at the epicentre of the global economy finds itself struggling to maintain its economic competitiveness, as two overseas rivals emerge to exploit their own local advantages and economies of scale. The struggle to maintain global power and dominance over the economic system places increasing strains upon the public finances of the state at its centre. A financial shock occurs, triggering a deep and lengthy recession across the capitalist world. The recession leads to the decline of previously successful ways of organising industry, but also opens up possibilities for new forms of production harnessing new technologies. As the new organisational forms emerge, so new skills, professions and values develop to harness and humanise them. In the decade that follows this economic slump, one particular model of the firm spreads with astonishing speed. It goes on to become the dominant productive force of the new century.

This scenario occurred between 1873 and 1903. The world power was Britain, the two competitors Germany and the USA, the financial shock the collapse of the Vienna stock exchange and the new technologies electricity, steel and petrochemicals.⁸ But it is the ascendant organisational model that concerns us here: the publicly owned business corporation.⁹ Britain's industrial dominance had rested heavily on one particular model of the firm, namely small-scale producers owned by families and industrial entrepreneurs, trading between each other and exploiting a global empire. Foreign rivals sought to compete collectively through forming cartels and trusts, but many of these were outlawed in the USA and Germany during the 1890s. Instead, both countries witnessed a vast wave of mergers, the result being the large, hierarchically organised, professionally managed firm with which we are familiar today. And in the USA, ownership of these firms was distributed amongst public shareholders.

Any firm is a cooperative of one sort or another. In the case of the publicly owned corporation, it is notionally a 'capitalist cooperative', a large number of individuals and institutions who pool their capital to take collective ownership of a firm and vote on its direction. This distributed ownership model, facilitated by stock markets, allows firms to grow at a speed and to a scale that was previously unimaginable, creating new possibilities for investment and production. It also requires a class of professionals who are delegated to take control of these vast bureaucracies on behalf of the owners. A hierarchical, deferential culture and ethos is also needed, through which these professionals can direct skills and labour in the most efficient manner possible.

To be successful, a model of the firm must not only be efficient, it must fit with the culture, politics and value system of the society in which it sits. The growth of the publicly owned corporation in the late nineteenth century is partly explicable in terms of the cost advantages that large hierarchies then possessed over Britain's decentralised market system. But it was also validated by a particular vision of cooperation, rooted in professionalism, a bureaucratic ethos, an expanding state, and excitement about the possibilities offered by far-reaching, vibrant financial markets.¹⁰

The evolutionary perspective

Over a century later, and immersed in our own economic crisis whose duration is still unclear, the question is whether this is still the model of economic cooperation that suits us. Is it necessarily the case that ownership of companies should be distributed across external shareholders to whom dividends are paid? Is the separation of ownership and control that was the cornerstone of twentieth-century corporate governance necessarily best? Does the subservience of employees to professional managers fit with the expectations of people in twenty-first century Britain? Are we content with the notion that firms are primarily accountable to stock markets? Should the government's attitude to industrial organisation only focus on how it affects the welfare of consumers? This chapter suggests that the answer to these questions may, in various circumstances and for various reasons, be 'no'.

As the nineteenth century example indicates, economic crises are complex events. They may be sparked by a slump in one particular market, but can only be thoroughly understood with a much broader perspective. Historians will note that the current crisis was initiated by defaults on American sub-prime mortgages in the run up to summer 2007, but will want to go further and understand the culture, psychology and politics that underpinned the perilous attitude to credit and the veneration of financial services that grew over the preceding decades.

Chris Freeman, an evolutionary economist, and Francisco Louca argued that each epoch of economic history is characterised by its own distinctive settlement between five 'semiautonomous' spheres of progress, namely science, technology, economics, politics and culture.¹¹ These spheres bleed into each other and take time to re-settle after moments of upheaval. Sometimes there is no discernible relationship between two separate crises, and yet when they coincide they amount to more than the sum of their parts – think of how the combination of Sir Fred Goodwin's pension and the MPs' expenses scandal seemed at the time like symptoms of a single disease.

Yet as the now-famous Rahm Emanuel line, 'you never want a serious crisis to go to waste', implies, there are positive aspects to crises. Just as a variety of cultural, moral and political factors may contribute to them, so their outcomes can represent cultural, moral and political progress. The economic slump of the 1970s was not caused by the cultural upheavals of the 1960s, but the economic model that emerged – privileging consumerism, individualism and rising female employment – incorporated a 1960s value system within it.¹² Understood literally, a 'crisis' is simply a turning point whose outcome can be good or bad. When rethinking the firm in the current climate, we have the opportunity to consider multiple ways in which existing structures fail to satisfy us, not only on an economic level, but also on a broader cultural and political one. The intention here is to outline four inter-related crises in which the dominant model of the firm is implicated.

#1: The banking crisis and the firm

The shareholder value creed was a particularly acute attempt to assert the power of financial capital over industry, making return on investment the single benchmark against which all of a firm's managerial, technological and employee actions could be measured. It is therefore heavily ironic that the model's shortcomings and the myths on which it rested were eventually exposed most graphically within the financial services sector itself.

If the model had worked correctly, the shareholders that owned Britain's banks would have defended the value of their equity by holding management accountable for their decisions, and in turn management would have held employees accountable for the dealings that were being done on the bank's behalf. It transpired from 2007 onwards that bankers had been engaged in actions that were destroying the value and long-term viability of banks, and that there were no adequate accountability mechanisms available – or being used – to discover or thwart this.

As Martin Wolf has pointed out, bankers in stock marketlisted banks have a strong incentive to take risks with unusually high stakes, especially where they are being remunerated with stock options.¹³ They share in all of the benefits of rapid growth, but are protected from the small risk of a meltdown by the principle of limited liability. And yet the TUC's research on the voting behaviour of institutional investors showed they presented no obstacles to the remuneration packages that led banks towards such high stakes strategies.¹⁴ RBS's ultimately disastrous £50 billion takeover of ABN Amro in 2007 went similarly unopposed. The discovery that many banks are considered 'too big to fail' only exacerbates the problem of individual incentives for risk heightening.

It was of course the hedge funds that correctly identified the huge risks to which the banks had become exposed, and began to short-sell them into oblivion. As controversial as this was, it represented a greater force for transparency than anything that was being imposed by the supposed 'owners' of the banks. All the short-sellers were doing was correctly identifying the stock market's delusions and exploiting them. As the financial journalist Chris Dillow has pointed out, the fact that hedge funds weathered the storm better than banks is an indication of their superior ownership and accountability structure, in which the professional investor's own wealth is often at stake in their decisions.¹⁵

Building societies have suffered due to their close relationship with the property market, but they were never liable to fail in the dramatic fashion of the shareholder-owned banks. It is not simply that regulation prevents them from becoming involved in the types of speculative activities that the banks were engaged in (at least not to the same extent), but that their ownership and corporate structure gives them a lower appetite for risk.¹⁶ As mutual societies, they exist for the core purpose of delivering a service to their members, not chasing a profit. Even if they were permitted to engage in risky investment banking activities, it is not clear that it would be in their interests. Equally, save for the demutualisation wave of the 1980s and 90s, these organisations are not constantly for sale in the same way as a stock-market-listed bank; indeed, ownership is not technically transferable at all.

Contrast this with the spate of huge takeovers performed by RBS, achieved thanks to share price inflation, which led it to become the fifth largest bank in the world, before the 2008 calamity of the largest reported loss in British corporate history.¹⁷ Such rapid growth would be unthinkable for a building society, but then so would such rapid decline. The fact that it is demutualised former building societies such as Northern Rock that have been amongst the worst victims of the financial crisis only heightens the tragedy of allowing Britain's financial sector to become dominated by this pursuit of shareholder value. Ultimately, shareholders – and of course the British taxpayer – were the victims of a creed that purported to put them in charge.

One important conclusion that can already be drawn from the financial crisis is that over-reliance on a single model of ownership and corporate governance in the financial sector increases systemic risk.¹⁸ The *Financial Times* was alert to this in an editorial as far back as 1999:

A pluralist approach to ownership is conducive to greater financial stability. With their contrasting capital structures, banks and building societies balance their risks and loan portfolios differently. Systemic risk is therefore reduced.¹⁹

The demutualisation wave that followed the 1986 deregulation was fuelled largely by greed on the part of those wishing to take control of the institutions involved and a shorttermist psychology amongst members who were encouraged to view them as an asset to be unlocked. The fact that value was locked into them can now, in retrospect, be viewed as a virtue and not a failing. As chapter 5 will outline, there is now a strong case for remutualising former building societies and finding ways of locking value into financial institutions once more. The argument made here is simply that the presumption that corporate assets are worthless unless they are tradable has been made to look extremely short sighted by the financial crisis. This lesson has implications beyond the financial sector for how we consider firms in general.

#2: The competitiveness crisis and the firm

The importance of housing and financial services to the British economy grew progressively throughout the 15-year boom that followed the 1992 recession. Financial and business services produced almost a third of UK GDP in 2006, more than twice the contribution of manufacturing.²⁰ Our economy is now more dependent on financial services than those of the USA, France, Germany or Japan, whereas in 1992 it was the least dependent of any of these nations.²¹

The property bubble, supported by speculation and a quantity of debt greater than the country's entire GDP, not only fuelled growth in the construction sector but unleashed a wave of consumption that increased the rate of economic growth further.²² The risks involved in this 'consumer Keynesianism' were a great deal more transparent and manageable than those being taken by the financial innovators in the City, but it is impossible to generate wealth on the basis of credit for ever, no matter how much credit there is around. At some point, Britain was going to have to confront the awkward questions of how, where and by whom its prosperity was to be generated in the twenty-first century.

In the wake of the financial crisis, Peter Mandelson has seized the opportunity to call for a new era of 'industrial activism', following an epoch in which the very idea of governments having a policy on production was treated as a taboo. As Mandelson explained, industrial activism

means being pragmatic about the ability of markets to enable companies and people to succeed in a rapidly changing global economy. Policy should be activist in the sense that it recognises that government can and must complement market dynamics to get the best outcomes for our society and economy.²³

Elsewhere, he has spoken of the need for government to consider the potential to support industry with its procurement decisions and for policies to put Britain at the forefront of a 'new green industrial revolution', based around low carbon technologies. Further sectors that have been identified as worth nurturing include hi-tech manufacturing, aerospace, automotive, biosciences, precision engineering and creative industries.

Yet the financial crisis continues to drag down the very industrial sectors that offer hope for the future. Sustainable energy companies are struggling to get the finance they need.²⁴ This is on top of Britain's persistent productivity lag, associated with comparatively low rates of business investment and private sector R&D.²⁵ With the credit market still badly damaged by the banking crisis, government is increasingly willing to step in to support the formation of new businesses. There is still further scope for publicly owned banks to be used in a more strategic fashion, to support the types of sectors with longterm viability. Government may resist the idea of 'picking winners', but it now recognises the need to help grow high productivity sectors.

A great deal is changing, not only in the assumed relationship between government and market, but also in the attitude of policy makers towards individual sectors of production. The government can no longer afford to be agnostic on how and where wealth is produced. Mandelson gave a striking indication of the shifts in political economy when he went so far as to outline this new activist agenda to the City.²⁶ Quite aside from the banks' dramatically self-destructive behaviour, the status of finance in our economy and society will have to alter to become less autonomous, less short-termist and more alert to industrial and political priorities. Government ministers have delivered this message repeatedly to the City, emphasising that both boards and investment funds need to become more expert in long-term wealth creation.²⁷

A tension between the immediate interests of financiers and those of producers lies at the heart of capitalism.28 Financial psychology and culture prioritise fast, monetary returns; productive psychology and culture prioritise longer-term processes of technological investment, implementation and innovation. Different varieties of capitalism involve different relative weightings between the two. In Peter Hall and David Soskice's famous dichotomy, 'liberal market economies' such as Britain and the US tend to grant greater power to finance over industry, while 'coordinated market economies' such as Germany tend offer more equal status to both.29 To say that Britain may have to pay greater attention to its productive sectors is not to argue that finance must become entirely subservient to industrialists nor that neo-liberalism is about to be undone with a return to 1970s planning. It is simply to point out that Britain's famed financial services may have done an excellent job of selfexpansion over the past 30 years, but have been less successful at supporting the long-term needs of industry and innovation.

All of this heightens the case for new financial, governance and ownership models, which place long-term productive purpose at their heart, and not just return on investment. Venture capital is not working in Britain, either in financial terms or on the terms of the innovators who depend on it. The argument against the short-termism of stock markets is well rehearsed, although it is important to remember their strengths in particular forms of growth and risk-taking. Part of the new industrial activism must be not only the search for greater sectoral diversity in the UK economy, but also greater diversity of ownership, financing and models of governance. Now is not the time to shift from a 'liberal market economy' to a 'coordinated market economy', but to ditch the assumption that Britain has or requires a single governing 'variety' of capitalism at all.

#3: The fiscal crisis and the firm

One further irony of the crisis is that after 30 years of neo-liberal governments promising and failing to reduce the size of the state, it is only with the demise of neo-liberalism that real shrinkage can now honestly be expected to occur, for better or worse. The fiscal and monetary stimuli that followed the banking rescue package may have averted a depression, but nobody would claim that this level of public borrowing and spending were sustainable macroeconomic policies for the long term. Few doubt that public spending will have to fall from 2011 onwards, perhaps by as much as 10 per cent. If this crisis presents a 'left wing moment' as some have claimed, it is being experienced under harsh 'right wing' fiscal conditions, although the argument for higher taxes may yet attract political support. As Demos Director Richard Reeves has argued, 'progressive austerity' must now be the goal, in which enduring progressive goals must somehow be pursued while spending less public money.30

What does this have to do with the organisation of the firm? At least two things. Firstly, in an era of smaller government, it becomes more important that non-state-based sources of security and community are safeguarded and encouraged. Prior to the twentieth century era of the social democratic state, there were various non-state vehicles developed for progressive purposes. The cooperative and mutual movements of the nineteenth century set about delivering services and economic security to their members. Republican thinkers such as Stewart White argue that this liberal-left tradition needs rediscovering and celebrating, for the self-government that it offers.³¹ White's argument is not framed in economic terms, but Britain's current fiscal situation adds to the urgency of reconsidering mutuality as a source of social and economic security.

This argument must not be used as a fig leaf for the state's abandonment of the poor. And yet it needs highlighting that if policy choices are framed only in terms of 'progressive state vs conservative market', the former is already under excessive strain, and greater nuance is required. A more progressive and pluralist model of market actors will help here.

Away from the state's core responsibilities to those most in need, progressive austerity *does* require cuts to be made, and it is here that organisational and ownership models are implicated in a second way. Public services sorely need to become more productive, regardless of who owns the delivery vehicle. Some arms of the state will need to be shrunk or sold. The politics of this will be complicated and painful. But again, the stark choice between the state and market is too simplistic, exaggerating the capacities of states to deliver public goods and under-estimating the latent potential of non-state actors to do so.

As will be detailed in chapter 2, public services are already being effectively delivered by various bodies not technically owned or controlled by the state, including Foundation Trust Hospitals, employee-owned health service suppliers and public service broadcasters. The Liberal Democrats have promoted such a model for the Post Office for some time. From the point of view of public service and public interest, there are important distinctions between profit-*maximising* firms, profit-*making* firms and not-for-profit enterprises. If the state is to shrink in certain areas, the structure and ownership of the firms that move into its place become matters of urgent political priority.

These arguments will be met with cynicism by some progressives. After all, the claim that public and private sector organisations are substitutable or identical has tended to mean only the former being modelled on or replaced by the latter. The New Public Management movement of the 1980s and the PFIs and public private partnerships of the 1990s were all about seeking to drive private interests, private methods and private profit making into the public sector. It has been 30 years of oneway traffic.

It is now time for the reverse to happen. If it is not solely the state's responsibility to uphold the public interest, or to correct 'market failures', we will need more private organisations that are imbued with public purpose or, in the jargon of economists, generate 'positive externalities'. The *principle* of a public private partnership or public interest company is a good one. But after an era of public services being remodelled around the private sector, it is now time to consider how private organisations might be designed with a greater sense of public service. Challenging our ingrained assumptions about the necessarily profit-maximising nature of private organisations is part of that.

#4: The moral crisis and the firm

Rising income and wealth inequality are inescapable features of recent British social history. According to one OECD study of 15 leading industrial nations, between 1979 and 1999 Britain's income inequality rose faster than any of the other nations studied, including the USA.³² Moreover, if the effects of taxation and redistribution are factored out, Britain's inequality is now at a higher level than in the USA, implying that markets and firms in the UK are producing unusually unequal outcomes in income.³³ The Institute for Fiscal Studies suggests that the Labour government has done a reasonable job in limiting these trends, most notably in its second term, but also finds that income inequality is now higher than at any point since at least 1961.³⁴

The story of wealth inequality is no less troubling. While one in five British households own no assets whatsoever, one sixth of the nation's assets are owned by just 3 per cent of households.³⁵ Things have been getting worse: in 1986 the bottom half of the population held 11 per cent of the liquid wealth of the nation, but just 20 years later this had fallen to 1 per cent.³⁶ For those who view a degree of asset ownership as a precondition of active, engaged citizenship, Britain faces a severe problem of a large dispossessed, marginalised minority.

A growing body of economic and psychological evidence pinpoints tangible effects this has on individuals – on their happiness, well-being and relationships and on community formation.³⁷ This is not only something that afflicts those at the bottom of society, although both absolute and relative poverty are associated with many of the worst of these effects, in crime, health, social capital and mental health. As Richard Layard and others have demonstrated, a culture of growing inequality affects the happiness of all members of society. Income is a surprisingly weak determinant of happiness, at least above around £25,000, while a culture of atomised aspiration and envy causes stress and community breakdown.³⁸

The question is posed once again – what does this have to do with the ownership or organisation of firms? A great deal. In general terms, these trends are associated with the rising status of finance capital in society and over firms. Studies have shown that income levels at the very top of the British labour market rise and fall in line with the stock markets.³⁹ There is also evidence showing that societies with more stock-market-led economies produce higher levels of inequality.⁴⁰ Meanwhile, inequality within firms has been rising, with average ratios of boss-toworker pay in the FTSE 100 at 66:1, based on salary alone.⁴¹ But if stock options and other incentives are included, this rises by half as much again, to 98:1.

The spiralling pay packages for those at the top of firms is therefore heavily associated with use of stock options to remunerate management, a key plank in the vision of the shareholder value-oriented firm. Externally owned firms, it was reasoned, suffer from what economists call a 'principal agent problem', where the split between those owning the asset and those controlling it leads the latter to exploit it in their own interests, not those of the owners. Use of stock options, whereby a staff member is given the right to purchase company shares at a fixed and typically meagre price at a time of their own choosing, is thought to align the two interests, giving managers a major incentive to grow the share value of the firm.

As a number of critics of shareholder value have argued, taken too literally this practice achieves the opposite of its intended outcome. Bosses who are granted large stock option packages are effectively being given shares at a knockdown price, inflating the share price, then selling for a vast personal profit. The option is a one-way bet: there is no risk attached to this reward, because if the share price drops they can simply decline to exercise the option. It is no surprise that some institutional investors are growing tired of these remuneration packages, which divert money away from both the business and the shareholders into the pockets of management. It is not clear that the beneficiaries are necessarily taking anything like the level of risk that might justify such extreme rewards.

The financial crisis exposed many such management practices to a public moral outcry. The scandal of Sir Fred Goodwin's RBS pension and the controversy over his successor's options-based £9.6 million annual pay package raised public and political concern about the naked unfairness at the heart of British firms. The financial sector may offer an acute example of the capacity for inequality that firms can unleash in the name of pleasing shareholders, yet these levels of income inequality are not restricted to finance. In 2007, for example, Giles Thorley, the boss of Punch Taverns, paid himself £11,276,000 or 1,148 times the average wage of his staff, to run a company that was heading towards significant losses.⁴²

Trends in the nature of work and production also appear to be exacerbating inequality in the UK. As the value added by 'human capital' – skills, expertise, know-how and creativity – rises, so the rewards to the highly educated begin to pull away from the rest. In certain circumstances, markets for knowledge and intangible assets produce 'winner take all' outcomes, whereby an ever-smaller minority of individuals or producers receive a rising share of wealth. The notion of the 'war for talent' propagated by McKinsey in the late 1990s pitted employers against each other in a zero sum game to acquire highly skilled, supposedly footloose individuals. Just as the shareholder value creed represented companies as constantly for sale, so the 'war for talent' rhetoric implied the same of sought-after skills. The trends exhibited by Premiership football clubs and players – zero loyalty, maximum greed – have been played out in professional services firms and creative industries around the world.

Firms in the 'coordinated market economies' of Japan and Germany do things very differently. Management personnel tend to be promoted from within the firm, and the overall ratio of pay between those at the top and those at the bottom is far lower. Economist Ronald Dore has suggested that where managers have worked their way up and seen the firm's social infrastructure from other perspectives, they identify more closely with their staff and the long-term interests of the firm, which constrains their own personal financial ambitions.⁴³

If there is a positive outcome to the moral crisis of British capitalism, it will be the restoration of some stigmatisation of spiralling inequality within firms. As for the use of stock options and other forms of profit-sharing, there is a great deal to be said in their favour, as chapter 3 will explore. Equity-based remuneration could potentially play a role in alleviating inequality, not only in income but in assets. But as the pay ratios cited above testify, it is presently being used as a mechanism to greatly increase the amount of inequality occurring within companies. The promises of financial participation and ownership have only been delivered on to a small fraction of the British public.

2 In praise of pluralism

Ownership is an inherently ambiguous concept. Individuals may feel some sense of ownership towards things that are not legally theirs, such as a local pub, just as they may feel very little sense of ownership towards things that technically belong to them, such as a buy-to-let property. Many philosophers and psychologists have begun their investigations of ownership from the recognition that people feel their body is 'theirs', and from this there stem various other forms of ownership over tangible and intangible goods.⁴⁴ A single concept conceals a wide variety of different freedoms, obligations, legal mechanisms, assumptions, norms, institutions and embedded relationships.⁴⁵ It is simplistic in the extreme to view ownership only as a binary, 'mine or yours?' issue.

Yet a prevalent view of firm ownership represents things in just such simplistic terms. The OECD argues that 'ownership consists of two main components: the right to income and the right to control'.⁴⁶ By this account, if something is 'mine' I am entitled to control it as I wish and to benefit from it financially. Applied to firms, this then produces the vision discussed in the previous chapter, that they exist to be exploited and disposed of according to the choices of their shareholders.

What the previous chapter sought to show was that this vision has failed in a number of respects, on its own terms as much as the public's. It did so because it not only ignored the ambiguous character of ownership in general, it ignored the complex realities of firms as social, economic and political entities, embedded in particular cultures. Quite evidently, companies are not assets that *can* be financially exploited and controlled like other forms of property, at least not without severely deleterious consequences for themselves and for society. We need to rediscover and celebrate the plurality of ownership

options, each of which represents a different way of combining economic, social and political priorities. These can coincide in the same economic system, and sometimes in the same hybrid organisational entity.

This chapter intends to do two things. Firstly, it describes *what firms are*, in explicitly political terms. Firms, it is argued, can only truly be understood in terms of the power relations that produce and sustain them. They are intangible entities consisting chiefly of intangible assets, and therefore cannot be fully owned, but only governed in some way or other. Secondly, it outlines alternative forms of organisation, in which power is more dispersed amongst multiple stakeholders, with equity investors either absent, better engaged or less dominant. This includes various types of mutual organisation, some of which are profitmaking, and various emerging organisational forms that could expand under the next phase of capitalism.

The politics of the firm

Let's start with the simple assumption that companies *do* exist for the purpose of serving the economic interests of their legal owners. Ensuring that companies are organised around this principle is, after all, the central task of most corporate governance models. Where firms are new, small or underdeveloped, the problem is a relatively simple one. A firm is likely to be initially founded, managed and owned by a single individual or family, such that ownership and control are combined in a single set of hands. As the previous chapter noted, this was the vehicle for Britain's early industrial leadership, and it continues to make a significant economic contribution today. According to two recent studies, family-owned businesses account for close to 60 per cent of world GDP and 31 per cent of UK GDP.⁴⁷

Many of these firms may not genuinely be considered 'capitalist', in the sense that their activities may not be serving the interests of 'capital' in any clear way. A shop that is handed down from one generation to the next, without any debt attached to it, can be run as the shopkeeper sees fit, with any profits going to the shopkeeper or being reinvested in the business. If the owner-manager decides that he wants to expand the business, he has the choice of accumulating his own reserves to do this or going to a bank. The debt finance offered by a bank serves as a capitalist means of expanding in size, potentially increasing profits, but without relinquishing ownership or control of the business, so long as the repayments on the debt are maintained.⁴⁸ More importantly, it is a basis on which businesses can be launched in the first place.

This type of scenario fits quite easily with a simple legaleconomic vision of ownership and control. The right of the individual or family to control and extract revenue from their assets, such as property and equipment, is not hard to defend. But at the same time, this comparatively simple model of the firm sits uneasily with the presumption that ownership is a narrowly economic phenomenon. If the founder or inheritor of a business chooses to retain complete ownership and control of the firm, this may be because it is highly profitable; but it is just as likely to be because they view it as having a more complex value and set of values than can be captured in its market worth. On the other hand, they may choose to split 'ownership' and 'control' by appointing a management team, recognising that others are better qualified to handle the assets in question. But then the concept of ownership starts to fracture.

This is where intangible assets come in, which leads unavoidably to questions of power. The individual or family that builds a successful business is really accumulating a set of intangible assets – most notably a brand name, a reputation with customers, a stock of knowledge, access to capital and a suitably trained workforce. The company's market 'value' might account approximately for these factors, but they are difficult things to price because they are not really *owned*. As Charles Leadbeater puts it, 'knowledge resides in people. People cannot be owned. Therefore companies do not own their most valuable assets.'⁴⁹ Investing in, developing and harnessing these assets requires complex social and political arrangements.

The most famous theory of the firm presents it as a political response to a particular economic problem. Writing in 1937,

Ronald Coase argued that firms arise to the extent that they do in order to overcome some of the inevitable inefficiencies involved in coordinating labour and skills via markets.⁵⁰ Tangible goods can be owned and distributed via markets with comparative efficiency. Labour, on the other hand, can only be *hired* and has the unusual economic characteristic that it can *speak*. Contracts must therefore be far more complex, covering far more possible contingencies, which raises the 'transaction costs' involved in negotiating and fixing them.

It is far more efficient, Coase reasoned, to organise people in hierarchies, whereby an individual is paid for a fixed period of time (a wage), but without the contract specifying *exactly* what they must do in return. Instead of either hiring on a task-by-task basis or specifying each task in advance (each of which is impracticably inefficient), the institution of the firm grants the employer *authority* over the employee, who is then expected to obey them within the limits set down in the contract.

The intangible asset in this instance is social and human capital. But other intangible assets also create problems that invite political responses. A brand and its associated reputation need protecting in various ways. Industries that rely heavily on intellectual property (IP) require firms to defend this 'property' in various ways, such as copyright, trade secrets, digital rights management and licensing. The lengths they go to in looking after IP – and the failures they encounter along the way – should alert us to quite how unlike conventional property it is. A software firm does not literally *own* the code it produces any more than it *owns* the employees it employs. But through various legal and technical mechanisms, it seeks to control who is able to gain access to it.

In what Jeremy Rifkin has dubbed the 'age of access', assets are not valuable because they are legally owned by us but because they are *available* to us.⁵¹ In this respect, a loyal customer base is an asset that has to be strategically nurtured and defended. This is supposedly achieved with the customer's consent, thanks to good service and advertising, yet with the rise of service contracts and advertising that targets the unconscious, firms have opportunities to constrain customers as well. The exertion of power over others is central to how capitalism operates, and much of what passes under the category of 'ownership' is in fact the freedom and power to dominate others.

A political vision of the firm is troubling inasmuch as it challenges the sunnier liberal economic view of individuals freely contracting with one another as they choose. The state is an accomplice here, providing the employment and intellectual property law that stands behind the orders issued by businesses. The French sociologist Jacques Donzelot identifies a crucial moment in the development of French capitalism as occurring in the mid-nineteenth century, when the French government granted factory owners the effective power to police and discipline their labourers during the working day.⁵² Of equal importance to this day, the state has to ensure that individuals are educated in a way that is compatible with the needs and ethos of employers.

In fact it is only thanks to a particular legal fiction that a firm is able to appear as an entity separate from its founders, creditors, managers and employees in the first place. The principle of limited liability, which emerged in the midnineteenth century, designates a firm as an independent entity that can receive investment and be liable for risks. But it is legally distinguishable from the people who own shares in it. This fiction is necessary to protect shareholders from being liable for its debts in case of bankruptcy, but it has had the bizarre side effect of creating a view that an association of people can be treated like an item of physical property.

This metaphor of association-as-property has created a second form of power relation, as discussed in chapter 1. The hierarchically organised firm becomes subservient to the interests of external asset holders or purchasers, which itself exacerbates the internal hierarchy of management over staff. The original justification for equity investment is that it enables firms to externalise risks around a larger number of people, and thereby to pursue more innovative, higher growth strategies. The firm gains access to a far larger pool of capital, while the public get to share directly in capitalism's remarkable capacity for wealth creation. The PLC model originally represented a particular accommodation between the needs of the public and those of industry. As we will touch on in the next chapter, stock markets have long held out the political promise of an inclusive, assetowning democracy.

With the rise of institutional investors and the pursuit of shareholder value, democratisation of corporate ownership has turned to centralisation, albeit in the name of several million pension and insurance policy holders. Only 15 per cent of UK shares are directly held by individuals today, down from over half in 1963.⁵³ Over the same period, institutional share ownership has risen from 25 per cent to 70 per cent.⁵⁴ The simple concept of ownership, that makes sense in the case of a shop-keeper's relationship to his shop, is transported into an entirely alien landscape of vast corporations, which require complex chains of hierarchy and accountability to be controlled. As John Kay has argued, a leap of logic occurs, in which legal ownership of a company share (the right to control and benefit from a slice of risk capital) is equated to ownership of the company itself (the right to dominate a complex socio-political entity).⁵⁵

Even more recently, a range of other 'owners', intermediaries and financial voices has emerged, in the hope of influencing and dominating firms further. Each of these relates to management in different ways, over different time horizons, from the pension holder who saves over several decades, to the private equity investor who engages with a firm over a few years, to the pension fund manager following quarterly reports, down to the hedge fund manager seeking to exploit sudden, short-term movements in stock prices. The think tank Tomorrow's Company presents an impressive examination of the full spectrum of company owners, evaluating the different forms in terms of the 'stewardship' they offer a company. As it argues, the rise of institutional investors and private equity creates additional distance between the suppliers of capital and companies themselves. As the report *Tomorrow's Owners* puts it:

With every new intermediary in the value chain that links savers to investment performance, there is an inevitable erosion of the sense of stewardship. In the global investment landscape, the value chains are now heavy with intermediaries, some of whose motives are unclear and some of whose activities can be destructive to long-term wealth creation and sustainability.⁵⁶

The concept of 'stewardship' helps to differentiate between different forms of equity capital, in terms of its sympathy and patience with businesses. Yet it still presents the firm primarily as a simple object with a legal owner, then examines the nature of the relationship between the two. This misses the various social, psychological and political dimensions of ownership, which enable a firm to cohere as an entity at all. The variety of ownership forms goes further still, once one considers the plurality of intangible assets and governance structures that characterise a firm.

Varieties of ownership

The first question is whether dominant networks of ownership and investment can be influenced to achieve this greater pluralism. Ethical investment funds, in which savings are only put into equities of companies with good environmental and social records, are already popular in Britain. Triple bottom line accounting, where companies report on their environmental and social performance alongside their financial performance, is voluntarily practised by firms seeking to create publicity around their ethical principles. These sorts of codes of conduct work with the grain of existing investment and accounting practices, seeking to subtly alter management incentives so that nonfinancial goals are pursued in addition to profit.

Some would like to see the democratic potential of stock markets exploited more fully. Institutional investment vehicles have considerable power over companies, and could be harnessed to serve the broader social and political interests of those whose money they are responsible for. In *The New Capitalists*, Stephen Davis, Jon Lukomnik and David Pitt-Watson argue for a new era in which the broader, longer-term social and economic interests of pension-holders are transmitted into the governance of companies. They believe that 'institutional owners accountable to their savers push corporations toward sustainable prosperity through responsible management^{2,57} Similarly, Tomorrow's Company seek greater 'stewardship' from external owners, in which they take a more active, long-term role in supporting successful companies.

If the civic potential of private pension-holding can be exploited, there is no limit to what political ends can be pursued in this way, at least where a large constituency of people has an interest. While the firm may remain a hierarchical entity, control over it can be democratised via the political power of 'citizensavers'. Trade unions are alert to the potential that their members' private pensions could exert over capitalism.⁵⁸ Meanwhile a number of projects, such as Enhanced Analytics Initiative, seek to help fund managers factor in the non-financial 'externalities' associated with their investments.

It should be noted here that there are other types of shareholder that are neither the imagined 'individual shareholder' that held sway during the 1960s nor the 'institutional investor' that dominates today. To be a shareholder is also to be a member with a vote, which carries intrinsic democratic value, though there are also varieties of share, as described in Table 1. A founder may retain a large share of a quoted company, with a 20 per cent block-holding representing a 'controlling interest'. Some individuals may choose to own shares in a company simply because they are closely affiliated with it in some other way. As we will discuss in the next chapter, share ownership by employees has an appeal, helping to align the interests of management and employees. However, while it may be considered desirable to engage more people in share ownership per se, this does not address the issue here of how to generate greater plurality in how firms are financed and governed.

While institutional investment via the stock market has risen over the past 30 years, it has been challenged over the last decade by a rival form – private equity. These funds buy large shares of a company directly, not mediated by the stock market, giving the buyer certain powers over how the company is run. This may be to help the company grow rapidly (as with venture capital) or it may be because it appears to be under-performing

Ordinary shares	Have no special rights or restrictions. They have the potential to give the highest financial gains, but also have the highest risk. Ordinary shareholders are the last to be paid if the company is wound up.	
Preference shares	Typically carry a right that gives the holder preferential treatment when annual dividends are distributed to shareholders. Shares in this category have a fixed value, which means that a shareholder would not benefit from an increase in the business's profits. However, usually they have rights to their dividend ahead of ordinary shareholders if the business is in trouble. Where a business is wound up, they are likely to be repaid the par or nominal value of shares ahead of ordinary shareholders.	
Cumulative preference shares	Give holders the right that, if a dividend cannot be paid one year, it will be carried forward to successive years. Dividends on cumulative preference shares must be paid, despite the earning levels of the business.	
Redeemable shares	Come with an agreement that the company can buy them back at a future date - this can be at a fixed date or at the choice of the business. A company cannot issue only redeemable shares.	

Table 1 Varieties of shares issued

and has greater potential for profit. Some private equity funds are themselves listed on the stock market, such as 3i. Of the largest 100 British companies not listed on the stock market, 39 have private equity investment, more than half of which had previously been listed.⁵⁹ The growth in private equity in the UK means that it now represents a significant challenge to the dominant, quoted company model of the firm: some three million people, or 16 per cent of the private sector workforce, now work for firms controlled by private equity.⁶⁰ In the UK, return on investment for conventional private equity funds is 11.8 per cent a year, compared with 4.5 per cent for venture capital.⁶¹ Evidently creating value in new ventures is a much greater challenge than extracting it from established ones. However, effective and patient venture capital is necessary to address the competitiveness challenges discussed in the previous chapter.

Could private equity be put to purposes other than maximising return on investment? Certainly private equity funds differ in their specifications of how high that return should be and over what time horizon it should arise. Tomorrow's Company identify that private equity is 'potentially the ultimate in shareholder stewardship' - but the keyword there is 'potentially'.62 Funds which pursue a vision of sustainable, prosperous organisations with long-term investment horizons represent a very different form of capitalism than those which spread themselves thinly over hundreds of start-ups or buy large companies in order to asset-strip them. Britain currently has more venture capital as a proportion of GDP than the USA, Germany or France, but it is not clear how well it is supporting long-term business generation.63 The average period of investment of these funds is currently four years, but the question is what sort of governance and expertise they contribute during that time.⁶⁴ Too often equity is provided on the condition of substantial control, but only exercised with that limited time horizon in mind.

How far private equity can be used to pursue non-financial goals depends on the investors and companies concerned. Community Interest Companies were established in 2005 as a new organisational form, which existed to serve some community purpose, but which could still sell shares and return a dividend, capped at 5 per cent. Assets are locked in to the company, to prevent it being bought and sold for greater profit. Helping social enterprises and not-for-profit organisations gain access to equity finance, with the greater growth potential that it holds, has been a long-standing objective for many working in this area. Various 'social venture capital' funds now exist, such as Bridges Ventures and Triodos Opportunities Fund, which exist to help profit-making social enterprises expand. There is even an argument for establishing a separate stock exchange for social enterprises.⁶⁵ We will return to this issue as a policy challenge in chapter 5.

Yet a far more radical break from the conventional understanding of company ownership, be it public or private equity, is where neither the company nor its shares can be bought or sold at all. This is a form of organisation that can be loosely referred to as a mutual, and it challenges many contemporary assumptions about the meaning and value of ownership. Mutuals are societies, trusts or cooperatives that own assets and are run on behalf of their members. Table 2 lists the dominant types of mutual in the UK, along with their economic and membership sizes.⁶⁶ The rules governing these associations differ depending on their goals, as the next section will explore. But one thing that they hold in common is that, while members are the owners of these organisations, they cannot sell this member-ownership, save where there is some strategic demutualisation, as occurred with ten of the largest building societies during the 1980s and 90s.

The obvious limitation of this form of ownership is that expansion occurs almost entirely through bank loans or retained earnings. Yet hybrid options are feasible, that give some access to equity finance, but within the strictures of a mutual structure. Building societies now offer Permanent Interest Bearing Shares (PIBS), which are tradable on the stock exchange but which pay out interest, rather than a share of profits. Equally, a portion of a company's equity may be held in trust, while the rest may be directly owned by shareholders. Hybrid models of cooperatives are emerging, in which 'investor-members' get to supply equity capital, which appreciates over time and can be sold internally, but without granting increased voting rights.

The Baxi Partnership offers a further hybrid option, dedicated to creating and supporting employee-owned companies. Baxi manages a £20 million endowment, which it uses to finance employee buy-outs and employee-owned startups. It provides advice for companies and entrepreneurs establishing themselves as employee-owned, but does not seek to

Sector	Number	Members	Employees	Revenue (£s)	Assets (£s)
Building Societies	55	23,038,000	51,578	4,700,000,000	360,000,000,000
Friendly Societies	200	6,000,000	4,600	2,271,005,984	17,500,000,000
Mutual Insurers	33	10,564,159	20,102	3,458,468,000	77,000.000,000
Co-operatives	2,753	9,008,221	159,434	20,123,892,555	6,518,625,537
Credit Unions	480	733,093	1,079	30,909,000	558,000,000
Housing Associations	1,879	5,328,335	143,654	10,800,000,000	52,477,000
Leisure Trusts	120	*	26,000	620,000,000	
Football Supporter Trusts	155	115,941	11	5,211,935	5,297,546
GP Co-ops and Mutuals	40	×	8,000	150,000,000	
Cooperative Trust Schools	-	*	140	6,400,000	
Clubs and Societies	10,000	9,000,000	25,000	463,000,000	219,915,731
Employee-Owned Businesses	200	¥	110,000	25,000,000,000	
	80	1,120,000		16,162,000,000	14,562,000,000
Total	16,005	64,907,749	844,208	83,790,887,474	476,416,315,814

Breakdown of mutual organisations in the UK67

Table 2

influence their direction or investment decisions in the way that conventional venture capitalists might. Viable, profit-making companies receive equity investments from Baxi of greater than 50 per cent share, but the value of that share does not grow. Instead, a return on the investment (equal to a proportion of the company's value added) is paid back to Baxi, and the equity is gradually bought back by the employees over time. This way the £20 million endowment is preserved and grown.

The membership and governance of any mutual depends on what it has been established to do. One thing that is clear is that none of the organisations categorised in Table 2 exists purely to make a profit, although some of them – including employee-owned businesses – do return a dividend. But as we have seen, it is relatively difficult to view *any* organisation with *any* ownership structure as existing purely to make a profit, and to do so is destructive. Despite the rhetoric and pursuit of shareholder value, the inescapable fact remains that a company consists of hierarchies, relationships and intangible assets that evade the narrow, legal-economic definition of ownership. This helps to explain why treating companies as tangible property, and buying and selling them as such, produces such poor economic results – it fails to take seriously the curious nature of the firm as a social, political entity.⁶⁸

A form of ownership that cannot be sold may frustrate some of our short-term economic impulses, and flies in the face of financial orthodoxy, but sits much more closely with the ambiguities of ownership and firms discussed in this chapter. As much as we may want to have private, transferable, legal ownership of intangible assets, it is not possible. The psychological dimension of ownership therefore becomes all the more important where goods are shared, relational and intangible. In this respect, mutual-type ownership structures are a more honest, and therefore more sustainable, basis on which to cooperate and organise, not only in pursuit of public and charitable goals, but in pursuit of private ones too. There is no automatic reason why firms owned in this way should be less hierarchical or less profitable than firms owned by shareholders. But the absence of financial pressures and the recognition of an organisation's social complexity creates tremendous opportunities for alternative purposes and freedoms to arise. These are facilitated and safeguarded by governance structures.

Varieties of purpose, varieties of governance

Remove the demands of financial markets, and an exciting question arises: what is a company for? As we have already discussed, this is an unavoidably political question. In the first instance, a firm is a hierarchical entity that exists to exert an element of control over the various intangible assets that it depends upon to sustain itself as a producer or service supplier. This includes knowledge, employees, IP, creditors, suppliers and loyal customers. But in whose interests?

The motivation behind founding and growing firms is more complex than the pursuit of profit alone, although some reward is expected for the risks involved. We will see examples in the next chapter of entrepreneurs who specifically sought to create something more valuable and enduring than a mere money-making machine. Especially in more 'coordinated market economies', this vocational approach to business can often be found in very large companies, where senior management have been promoted from within and have an intimate understanding of the processes, technologies and traditions that enable the firm to pursue its institutional purpose. One of the many risks taken in the run up to the financial crisis was when bankers abandoned a vocational attachment to the core purpose of a bank in favour of financial 'innovation'.

Firms may require a degree of hierarchy in order to coordinate their activities and to avoid the 'transaction costs' of constantly resorting to market contracts. But there are two ways in which the power of managers can and should be challenged.

Firstly, there is a political argument for installing democratic structures within firms, in order to ensure that they serve the interests of their constituents. This typically involves the creation of two boards, one containing managers or executive directors who are responsible for day-to-day decisions, the other containing trustees and elected representatives who are responsible for upholding the interests of the organisation's members. In this respect, the formal separation of 'ownership' and 'control' remains in place, at least inasmuch as the 'owners' (in this case whoever the members are) do not directly control the organisation. However, the 'owners' have to be consulted periodically and on issues concerning the long-term direction of the firm, and there are varying degrees of overlap between the two boards.

This is a renowned feature of German corporate governance, where firms rely more heavily on bank loans than on equity finance. The non-executive board will typically feature representatives of banks, employees, employers associations and unions. In the case of cooperatives, the members are those people who choose to trade with the cooperative, who purchase an ownership stake for a nominal fee, then elect representatives to the board. Where public services are run in this way, the nonexecutive board will contain representatives of the users of those services, as occurs with Foundation Trust Hospitals. There are a number of precedents of other public services being run by mutuals, with 600 cooperative schools in Spain dating back to the 1930s, and there is scope for expansion in the UK.69 Employee-owned public service providers, such as Greenwich Leisure and Central Surrey Health, have also emerged in recent years.

The design of governance follows from the central purpose of the organisation in question. Yet it does not help to view this in a very strict sense – to install a narrowly-defined social or economic purpose in place of profit-maximisation may be to replace one simplification with another. The virtue of democratic governance structures is that they enable debate, ambiguity of purpose and plurality of voices, at the very top of an organisation. Chains of accountability are important, but it is dangerous to view them as absolute. One of the great virtues of cooperative governance structures is not that they enable companies to be run *purely* in the interests of consumers or *purely* in the interests of employees, but that a variety of ethical and social factors can be fed into decision making, for which the membership may not be technically responsible. Environmental concerns would be one such example. The second way in which management power can and should be challenged concerns the cultural and economic aspects of the intangible assets that they are overseeing. The culture of advanced capitalism has changed substantially since Coase's time, and the model of the despotic, hierarchical firm no longer works to the same extent. The decline of manufacturing and Taylorist management means that flexible production, creativity, human capital and collaborative working are now at the heart of our economy. This is partly about the shift to a 'post-Fordist' economy built around service industries and fast-changing consumer desires, which are satisfied symbolically as much as objectively. But it needs acknowledging that it has been thrust upon firms by employees as well. As Paul Miller and Paul Skidmore found in a previous Demos report:

Employees want more human organisations with greater autonomy and flexibility. They want an experience of work that is aligned with their values. They want a workplace forged in the image of their identities, not a workplace that tries to define them. They want organisations that can let go, and grant them a greater say in how things are run.⁷⁰

Employees who are listened to in this way may be a fortunate minority, but the point still stands that the expressive, consumer ethos that has transformed our economy in recent years also penetrates the workplace.

There is a robust economic case for granting employees greater autonomy in the workplace, for instance through increasing their task discretion.⁷¹ It is estimated that one-third of British productivity growth and half of labour productivity growth between 2000 and 2005 was due to investment in various intangible assets, rooted in people.⁷² Employees hold knowledge and information that is crucial to the economic success of their firm, and this cannot be utilised effectively if their relationship to management is purely one of subservience. The political character of the firm has had to change out of economic necessity, and will have to change further if Britain's enduring productivity deficit is to be eliminated. An empowered workforce is now a productive asset. But there are additional reasons to oppose excessive workplace hierarchy that are not reducible to the productive gains that can be made. There are moral and political arguments for granting employees power over their working practices. Republican political theorists criticise certain elements of capitalism for the lack of control and meaning that it grants workers. Thinkers such as Nien-He Hsieh and Stuart White turn to democratic governance structures such as cooperatives as a means of redressing this. While there are tensions between the needs of capital and the positive freedoms of the workforce, Hsieh believes that 'it is possible to imagine work that meets the criteria of meaningful work within the context of the employment relationship'.⁷³ Political empowerment matters at the level of the individual and the task, as well as at the level of the firm and society.

This links to a moral argument about the nature of work and the importance of self-expression. The sociologist Richard Sennett has written of how the modern workplace destroys the capacity for human beings to tell a coherent story about their lives and to develop the capacity for craft.⁷⁴ In this respect, Sennett holds the post-bureaucratic, flexible workplace partly culpable for preventing individuals from growing expert in particular crafts or identifying with their work over time. It should be acknowledged that an 'engaged' and consulted workforce *may* opt for repetition, security and gradual skill development, and not endless flexibility and change. Another way of framing Sennett's argument would be that contemporary workers don't feel adequate ownership over their work – the psychological dimension of ownership that evades economic or legal categories.

The psychology of work and ownership rebounds back on measured outcomes for employers and society. According to the World Health Organisation, stress at work is associated with a 50 per cent excess risk of heart disease and a range of other mental and physical health risks.⁷⁵ Worryingly, the number of people in Britain who claim to be working 'very hard' or 'under a great deal of tension' has been rising steadily since the 1980s.⁷⁶ The risk with focusing on 'employee engagement' in purely economic terms is that some workers may be more autonomous,

53

but also more stressed and unable to distance themselves from the workplace when they want to, as some evidence has confirmed.⁷⁷ One recent study has actually shown a *decline* in the number of employees in the UK who feel that they have a great deal of influence over their work.⁷⁸

The opportunity offered by firms that do not identify as profit maximising (which is not the same thing as profit making) is that work may be more rewarding, meaningful and autonomous. If one thinks of the various intangible assets that are organised to produce a firm, each of these potentially carries intrinsic value, and not only instrumental economic value. Customer service, creativity, cooperation between colleagues, innovation, investment in knowledge, public service and much else are all phenomena that *can* be valued intrinsically, rather than mobilised as a means to generate profit. A society that wants more of these things - for whatever reason - needs to produce the structures that will produce and defend them. Economists tend to describe such side effects as 'positive externalities', add-ons to the more primary mechanisms of exchange and competition. Why not consider them as goods in and of themselves? This may involve founding cooperatives, trusts and societies that exist for an explicit purpose. But in the first instance it depends on reducing or redirecting the power of financial markets over firms.

The future of cooperation?

The evolutionary perspective on the firm sees it as drawing on and contributing to its own particular social, cultural and technological contexts. As these contexts change over time, so structures of cooperation and production change with them. If all firms are cooperatives of one sort or another, albeit with different ways of recognising this, it is worth considering the emerging forms of cooperation in 2009, within which may lie the seeds of future models for the firm or alternatives to it.

Over recent years, the internet has facilitated an entirely new solution to Coase's problem. Rather than organise assets within a hierarchical structure, with people hired on a regular wage, they can be brought together on a project-by-project basis, dispersing in between. Firms have already learnt the art of 'just in time production', but we are now witnessing the rise of the 'just in time producer'.

This was initially witnessed in the early 1980s with the birth of open source software projects. The rise of 'social media' and social networking sites over the last decade has meant that work, information-sharing, creativity and creation of intangible assets can occur without people first being organised into cooperative units. Individuals can participate in these activities as it suits them, and others can contribute, join in or benefit as and when they please. As Clay Shirky says, we can now 'organise without organisations'.⁷⁹

The one condition on which this model of cooperation depends is that the whole notion of economic 'ownership' of intangible assets is abandoned once and for all. This does not mean that there is no control involved. On the contrary, open source software and open access cultural content depend on legally binding licences to restrict certain types of use, such as those that would privatise, commercialise or lock up the assets in question. For instance, a range of Creative Commons licences can be added to copyrighted material, to specify the precise ways in which it may or may not be used. One might say that where mutuals have their value *locked in* to them, open networked collaboration practices make sure that value remains locked out of any private hands or organisation. If most firms seek to defend their intangible assets as a 'private good', and mutual societies do so as a 'club good', then open networked collaboration does so as a 'public good'.

These networks are capable of replacing some of the functions of traditional firms. They enable information-sharing, community formation, low-budget R&D and so on, but without any management at all. There are some businesses that are directly threatened by this new model of cooperation, especially those businesses that rely on collecting and selling codified information. Many newspapers and magazines profited hugely from selling box advertising space, a market that has now been effectively replaced by free websites such as Gumtree. On the other hand, it may be that traditional organisations can *gain* from enabling open, networked association to flourish within them, as Peter Bradwell and Richard Reeves have argued.⁸⁰ The fact that these networks are not coordinated by prices or wages does not mean that they are not productive economic agents.

By the very nature of such associations, they are ill-suited to generate revenue, save through exploiting their public character and selling advertising space. For this reason, there are very few examples of such structures providing a basis for secure employment. Examples do exist of contingent labour being organised by new online networks, such as Amazon's Mechanical Turk and Slivers of Time, which mediate between potential employers and potential employees, to provide chunks of work split into individual tasks. School of Everything mediates between potential teachers and potential learners, taking education and training outside of the school, university or business. What these examples indicate is that, in Coase's terms, the 'transaction costs' of buying and investing in labour have fallen substantially, and we are consequently that much less reliant on hierarchies as a means of organising our productive assets.

This form of association discards the sense of membership that mutuals and orthodox firms rely on. Barriers to entry are virtually non-existent and the benefits cannot be contained, making them difficult vehicles for private investors to support. Where they have no fixed costs or wages to pay, this is not a problem. At present, 'organising without organisations' is presenting a very real problem to traditional firms which do have overheads, especially newspapers. But where open networks themselves have overheads, they are often dependent on charitable donations or some other third party support. Wikipedia and mySociety, for example, are founded as charities dependent on donations. Channel 4 has now established a fund, 4iP, to support more such ventures, some of which are revenue generating, but not all.

If mutuals take one step away from capitalism by abandoning profit-maximisation, then these networks take another step again, by abandoning the use of prices and wages. In this respect they often disrupt other firms more than they can replace them. But if open networked collaboration cannot quite replace the firm as the basic social unit of capitalism, does it have any potential to replace existing mechanisms for coordinating financial capital? Already there is a small-scale alternative to banks and building societies in the form of Zopa, which brings together lenders and borrowers online to bypass intermediaries. Individuals trust others to repay loans on the basis of independent credit ratings, achieving better rates of interest than banks or building societies are able to offer. Borrowers can access up to $\pounds_{15,000}$.

Applying a similar principle to equity investment is problematic, for the simple reason that a higher level of expertise is required. A peer-to-peer venture capital model can be imagined in which hundreds of users buy shares in a start-up not mediated by the stock market, but the viability of a new business is not something that can be easily understood by amateur investors or communicated via a simple rating system. On the other hand, it would appear that the internet significantly lowers the barriers to the creation of new consumer cooperatives or collectively-purchased assets. If participants in an online network all desire something, such as more renewable energy, then they ought in principle to be able to agree to pool their money as a form of equity finance.⁸¹ Abundance, a new venture coming out of the innovation unit of Imperial College Business School, Design London, promises to support exactly this, coordinating amateur investors to purchase wind-farms collectively and receive a return. Then there are smaller-scale examples of social media being used for collaborative purchasing. The mySociety website Pledgebank has been used in this way, for instance with the successful pledge 'I will donate \$20 to purchase a new energyefficient refrigerator [for a local social enterprise] but only if 49 other people will do the same.'

The problem these associations have is that they struggle to appoint decision makers or to overcome their defining feature, which is the freedom for people to exit and enter from one moment to the next. A mutual establishes a set of democratic rules that everyone then consents to. Power becomes centralised, but on the understanding that it is used within a constitutional structure. Open collaborative networks, on the other hand, have certain individuals who are more powerful than others (the highly connected, the initiators, the brokers) but little sense of legitimacy surrounding these individuals, and therefore little obligation for others to obey them. The bodies providing the 'constitutions' of the new associations tend to be private companies – Facebook, Google, Zopa etc – and their rules are currently established as they please. If a new *democratic* mutualism is to arise from any of this, and not merely a new informational mutualism, then this problem will have to be overcome.

3 A better model of the firm

Capitalism is always a curious mixture of liberalism and despotism. It grants freedom, equality and participation with one hand, while it imposes orders, hierarchy and inequality with the other. We barely notice that we live schizophrenic economic lives, in which half our waking hours are spent being indulged and obeyed as sovereign consumers, while the other half is spent being ordered around as employees. But in neither case are we treated as responsible adults. To live responsibly, both as consumers and as workers, would mean exercising economic freedom with a sense of the stakes involved. The liberal and the despotic elements of capitalism would be brought a little closer together, so that freedom was never untrammelled and power never unaccountable.

At the centre of an argument for employee ownership is a subtly different vision of how economic power should be organised and utilised. Managers possess a greater sense of accountability to the employees who own the firm, while employees acquire a greater sense of responsibility for the assets that they are tasked with using and developing. An alternative form of autonomy is at work, that is very different from that demanded by the investors and executives that control listed companies.

There is a rich tradition of employee ownership, much of which is distinctly British. Given the 'path dependency' of how national economies evolve, this matters. Workers' cooperatives have prospered in Britain since the nineteenth century. Between the 1930s and the 1960s, the Liberal party expended considerable energy defining a political economy that was neither capitalist nor socialist, but granted freedom through ownership.⁸² And use of employee share ownership plans is known to be more popular in English-speaking 'liberal market economies'.⁸³ In many cases, this tradition is the counter-movement against the more ruthless,

61

Table 3 Definitions of employee ownership forms⁸⁴

'Employee ownership'	Employees own more than 50 per cent of the shares, either directly or indirectly
'Employee share ownership'	Employees own some shares in their firm directly
'Workers' co-operative'	Employees own firm indirectly as members, not as shareholders, with rights to be represented at board and/or management level
'Co-ownership'	Employees hold substantial minority stake of more than 25 per cent

short-termist aspects of British capitalism, which explains part of its appeal here. But in the context outlined in chapter 1, there is a good case to normalise many parts of this agenda.

This chapter examines employee ownership in detail, including workers' cooperatives and other forms of financial participation such as employee share ownership. Table 3 offers clarification on how these terms are used, although there is considerable overlap between them. There is ample evidence that, structured well, employee ownership can create a uniquely virtuous circle between workplace democracy and productivity.

There is one crucial technical distinction that needs to be explained and stressed at the outset. This is the distinction between 'direct' and 'indirect' employee ownership. *Direct* employee ownership means employees purchasing or being given shares, a practice that governments have sought to encourage with tax incentives, as discussed in the next section. *Indirect* employee ownership means a firm's equity being placed in trust or other type of mutual society, which acts on behalf of all employees, and distributes profits amongst them. This distinction has very important implications for how risk and reward are distributed within firms. For example, the common critique that employee ownership increases the economic risk faced by staff, by making their savings and jobs dependent on a single entity (the 'eggs in one basket' argument), only applies to direct ownership. Firms such as the John Lewis Partnership, which operate with indirect employee ownership structures, do not require – or for that matter permit – staff to invest any of their own money at all.

A common assumption, even for some who promote workplace democracy, is that it is necessarily traded off against productivity. There may be extremes and certain industries where this is the case – extreme forms of workplace democracy may leave no scope for managerial discretion, while there is no doubt that productivity *can* in many industries be achieved through increased managerial domination. Any mutual organisation must somehow balance its commitment to member representation as a political good with a support for efficient management as an economic good. The most representative governance structures are not necessarily the most effective or efficient, but then the most effective and efficient are not necessarily the most fair. This problem is grappled with practically when mutuals and employee-owned firms set about establishing or re-establishing their constitutions, most tangibly in the balance of power between the board of directors and the board of member representatives or trustees.⁸⁵ The appeal of employee ownership is that the democratic and economic benefits of mutualism can start to reinforce one another. Let's look at these in turn.

The democratic gains

One political curiosity of employee ownership is that it has been claimed as a goal by both left and right. The left has viewed it as a cooperative, ethical alternative to exploitation by capital, while the right has viewed it as a means of diffusing industrial disputes and a way of bringing the interests of labour into line with those of management. But whether it be capital that becomes more amenable to labour or vice versa, the central point is that dispersing ownership and governance rights throughout the firm creates a greater sense of *legitimacy* about the decisions that are taken. This is partly because participation in decision making is higher, but also because the decisions that are made are more ethical, factoring in a broader set of interests.

The most prevalent form of employee ownership in the UK is employee share ownership, a form of direct ownership. At the absolute minimum, employee share ownership can occur without any significant governance implications at all. After all, an employee is free to buy stock in any company, including their own. However, successive governments have used tax incentives to increase employee share ownership, starting with Save As You Earn in 1980, followed by the Company Share Option Plan in 1995, then by the Share Incentive Plan in 2000. These are described in table 4. In Britain, around 20 per cent of workplaces currently use some type of employee share ownership scheme, covering 32 per cent of all individuals in employment.⁸⁶

There is an ideological component to these plans. The USA has a long tradition of viewing share ownership as an important component of its democracy, the twentieth century equivalent of land ownership. Starting with Louis Kelso in the 1950s, then picked up by Senator Russell Long in the 1970s, the case has been made for drawing all members of society into the stock market. As Jeff Gates, the heir to this tradition, argues, the problem with contemporary capitalism is that it 'is not designed to create capitalists but to finance capital'.³⁸ Capital ownership needs dispersing in the form of shares. The British Conservative government was sympathetic to this project over the 1980s and 90s.

But equally the 'liberal republican' tradition views broadening asset ownership as an important precondition of a participatory, empowered citizenry.⁸⁹ Rather than capital being taken into the hands of the state, where it retains much of its monopolistic capacity to dominate, it must be dispersed and made more subservient. In the case of worker cooperatives, such as the famous Mondragon retail and finance network in Spain, this occurs through 'labour hiring capital' rather than 'capital hiring labour'. The principle of profit-sharing is an additional liberal republican means of tempering the power of capital. As Stewart White points out, initially this concept rested on the notion that both labour and capital have an 'income right' –

Table 4 Varieties of tax-advantaged employee share schemes in the UK

	How it works	Uptake	
Save As You Earn (SAYE)	The company grants a three or five year share option, often at a price discount of up to 20 per cent, to employees who commit to save regular amounts of up to £250 per month. When the option reaches its term, the employee can decide whether to exercise the option, using the savings to pay for the option, or simply withdraw the savings if the option is not worth exercising.	The scheme is popular in public companies, providing a risk- free savings and investment opportunity for employees. Around 780 companies operate SAYE schemes, including two- thirds of FTSE 350 companies. ⁸⁷ However, the scheme has been criticised for not creating long- term employee ownership because many employees choose to sell shares immediately or soon after exercising their options.	
Company Share Option Plan (CSOP)	Like SAYE, except the company can choose which employees may benefit, there is no requirement for a savings contract, and the size of the option can be up to £30,000. Any gains made by the employee are taxed as capital gains, not as income, which would be the case with 'unapproved' types of option scheme.	Commonly used by public and private companies to incentivise management and executive level staff, though there is no reason why the scheme cannot be operated by companies for all employees. Over 10,000 companies operate a CSOP or its small company equivalent, the Enterprise Management Incentive (EMI) scheme.	
Share Incentive Plan (SIP)	A flexible scheme in which the company may award employees free shares worth up to £3,000 per annum and/or may invite employees to buy up to £1,500 of shares from their pre-tax earnings, often sweetened by the company with matching free shares on a ratio of up to 2:1. To claim all the tax benefits, the employee must usually hold the shares for five years.	Since its introduction by the Labour government in 2000, this has now overtaken SAYE as the most popular scheme, with around 940 schemes in operation.	

return on investment is capital's 'wage'.⁹⁰ Profits are therefore shared between capital and labour, recognising the equal contribution of each. This principle has long been observed in France, where profit-sharing has been common since the 1840s, and is currently a statutory obligation for all companies of more than 50 people. Liberal republicans may well commit to employee share ownership on this principle.

Within the firm itself, there is compelling evidence that employee share ownership produces greater commitment between employers and employees. Studies in the USA found that firms with high levels of employee share ownership also experienced higher levels of identification with the workplace and job satisfaction.⁹¹ Moreover, this appears to be relatively unaffected by the number of shares owned – the mere fact of being an employee-owner, rather than just an employee, has positive psychological implications. This is further proof that ownership is not something that can be easily reduced to the legal-economic rights to a dividend and to control.

On the other hand, one of the central findings of research on this topic is that behaviour does not change unless financial participation is accompanied by other forms of participation in decision making. From an economist's point of view, this is because employee share ownership is a form of collective profitsharing, and so requires a sense of the collective in order to have any effect. From a democratic and social point of view, this is a virtue, not a defect. Once there is greater identification with a workplace and colleagues, various other positive spillovers tend to occur. In a survey of employee-owned and co-owned businesses in the UK, it was found that 84 per cent of these companies view better employee relations as one of the benefits while 81 per cent view more ethical behaviour as a benefit.⁹²

Employee share ownership (ie direct ownership) does, nonetheless, have certain limitations as a basis for empowerment and democracy in the workplace, or in society for that matter. Firstly, as an opt-in, individualised scheme, it only includes those who feel able to afford it. Unsurprisingly, evidence shows that participation in such schemes increases with wages, hence it potentially adds asset inequality to income inequality.⁹³ In the extreme cases of executive stock options discussed in chapter 1, it can exacerbate inequality severely. A convincing argument can be made that low income individuals should put their savings anywhere but in their own firm, as this makes them doubly vulnerable to it failing.

Secondly, employee share ownership rarely amounts to a new form of governance, even if it does produce higher employee engagement. In this sense, it satisfies the economic need for greater autonomy and engagement at work (as evidence in the next section shows), but may not satisfy the political need. Direct ownership of shares, in contrast to mutuality, may not in itself represent any form of challenge to existing hierarchies, whether it be the power of management or the power of capital. Where employees own a significant amount of a company (25 per cent is the level at which it becomes defined as 'co-owned') then this changes. The US mechanism of an Employee Stock Ownership Plan exists to progressively increase the employee stake over time, covering all employees collectively, potentially to the point where the firm is entirely employee-owned. UK tax-incentivised schemes operate individually, and while they count as a form of 'shared capitalism' in the sense of collective performance-related pay, they are unlikely to produce an outcome in which the managers are genuinely accountable to employees.

There are alternative, more radical forms of employee ownership that produce an altogether different set of relationships. These rest on various forms of *indirect* ownership, and follow the mutual model discussed in the previous chapter. A firm is established either as a workers' cooperative, in which every employee buys an equal share for a nominal price, or is owned by an employee benefit trust on behalf of all employees. Employees become owners (otherwise known as 'partners' or 'members') when they join the firm, but they do not contribute capital. Equally, they are unable to extract their stake or increase it. Since they do not directly own any of the company, they do not take any shares when they leave. As with all mutual models, the assets are locked in to the firm.

As owners of the firm, employees receive a dividend payment, reflecting the profits made that year. They also have a right to representation at board level. In cooperatives, this is dictated by the cooperative constitution, while other employeeowned businesses tend to design their governance arrangements themselves. The trustees of an employee benefit trust represent a second board, to whom senior management are ultimately accountable. The trust will have a declared purpose, which will be to govern its assets in the best interests of employees. If employees believe the business is not being run in their interests as owners, they have real, statutory power to take their concerns to this non-executive board, who then have the right to refer this to management.

It is important to note that the interests of these employeeowners do not militate against profitable activities in any way – far from it. The task of the trustees is to ensure that managers are doing a good job, on behalf of the employee-owners, who will want to see profits. But the definition of what good management looks like is enriched and the employment relationship more balanced, quite simply because managers are accountable for the power they hold.

The civic and social virtues of employee-owned businesses are difficult to specify with hard evidence. Customers appear to appreciate the ethics of employee-owned firms, which is one of the most commonly cited benefits associated with the structure.⁹⁴ They have the additional 'democratic' virtue, at least in comparison to share ownership schemes, of including all members of a workplace. How profits and voice are distributed depends on the constitution that is being used. But the possibility of power and financial reward being retained by a minority at the very top becomes inconceivable. This is an alternative to the 'winner-takeall' model of the firm.

There is another reason to favour this model for those who view firms as part of the patchwork of civil society. A major area of growth in employee ownership over recent years has been where family-owned companies and partnerships face the problem of business succession following the retirement of the existing owners.⁹⁵ The dominant solutions to this problem – a sale to a competitor, a private equity buy-out or floatation on the stock market – may return a higher price to the sellers, but they

threaten the enduring social and cultural values that define a firm. For owners who view firms as more complex entities than a market valuation can reflect, shifting ownership into an employee benefit trust ensures that the integrity of the firm can survive over generations.

The political theorist Edmund Burke argued that society is more than a contract between its existing members – it is a contract between the living, the dead and the unborn. Citizens and governments need to see themselves as inheritors and protectors of a legacy. This is also one of the most compelling ethical reasons for adopting employee ownership via a trust: managers and employees pay heed to the fact that they are the beneficiaries of a legacy, which needs preserving beyond their own tenure. This is an obstacle not to innovation (as we will explore next) but to the short-termist churn of financial markets, which fail to price intangible assets appropriately and therefore cannot be relied on to preserve their value.

The productivity gains

The economic evidence on the effects of employee ownership is remarkable. An Employee Ownership Index has been produced by law firm Field Fisher Waterhouse since 1992, tracking listed companies with significant amounts of employee ownership. The index out-performs the FTSE by an average 10 per cent annually.⁹⁶ Treasury-commissioned research on the government's employee share ownership schemes found that, in the right circumstances, they delivered a 5 per cent productivity increase.⁹⁷ These findings are confirmed by a number of other studies, both in the UK and in the USA.⁹⁸

Governance and participation aspects are crucial here. One thing that all studies in this area stress is that share ownership (or, for that matter, other forms of financial participation) only has discernible productivity effects when it is combined with other forms of participation and employee engagement strategies. The issue scarcely needs stressing where employee benefit trusts or cooperatives are concerned, as these automatically introduce their own participative governance arrangements. For example, the productivity effects of granting share options to management is negligible, but increases markedly when these schemes are extended to all staff in a workplace.⁹⁹ This is not only a fillip for workplace democracy, but also a fairly damning indictment of shareholder value-oriented practices.

It is important to understand why these effects might arise, which will bring us back to questions of power and democratisation. It is clear that employee ownership brings higher levels of commitment from staff, manifest in lower levels of absenteeism and greater willingness to take on responsibility.100 It stretches the time horizons of employees as much as managers. A simple economic analysis of this would suggest that employees have a direct incentive to raise their performance if they know that they will share in the profits via the share dividend.¹⁰¹ This is not unlike the supposed case for paying executives in stock options, which is that it creates a direct incentive for them to increase profits and share price. But a simple economic explanation would also suggest that *shared* capitalism, that is any scheme in which pay is related to collective performance, would lead to free-rider problems, where individual employees reduce their effort in the knowledge that their behaviour alone will not affect their bonus. Yet research on this issue of 'shirking' under shared capitalism finds no such thing.102

So what is going on? The concept of 'high performance work systems' goes some way to explaining it: combining varieties of workplace participation is known to produce higher rates of innovation.¹⁰³ A workplace is a complex social and cultural ecology. Much of what is invested in and 'produced' in twenty-first century Britain is intangible, and its quality depends partly on the relationships it is embedded in and the subjective evaluations of those who pay for it. Coase saw hierarchies as the only alternative to markets in situations where the latter were inefficient. But hierarchy is a clumsy tool with which to manage the post-industrial workplace, where it is the effort, commitment and innovation of employees that create competitive advantage. Unhappy, unmotivated staff are a significant and costly problem for managers, especially where they voice this by leaving a company. Employee ownership, combined with elements of workplace democracy, offers a distinct response to these challenges that goes beyond anything that can be achieved through tweaking human resource management (HRM) practices. Firstly, it makes explicit the psychological contract between management and employees. Both parties to any employment relationship have founded reasons for suspicion – management suspect that employees are not contributing as much as they could, while employees suspect that management are not being as honest about their motives and strategies as they could be. HRM policies focus on improving the quality of this relationship, but rarely address the authenticity of employees' concern: that the company is being run in the interests of shareholders, not them.

Employee ownership and representation address this in a sincere fashion. Offering employees a constitutional right to share in profits and participate in decision making begins as a signal from management that they are committed to dialogue and collaborative working, not simply to the extraction of greater effort. As the case studies will show, management operates in a more transparent fashion when the employees not only have rights to representation, but also have a vested interest in the performance of the firm. Employees, meanwhile, have a stronger sense of psychological ownership over assets of which they are the legal owners and intangible assets to which they contribute and from which they benefit. Surveys show that employee-owners take greater care with their work, hold management to account for decisions, and have a higher propensity to share knowledge.¹⁰⁴

Profit-sharing in the form of an annual dividend payment performs a critical role, but not one that can be easily understood via the narrow economic concept of 'incentives'. The dividend payment signals something about the power structures at work in the firm and the commitment of employees and management to it. Receipt of an annual dividend is something that management and employees have in common. The fact that it covers all staff (at least where indirect ownership is used) makes it a democratising force, in marked contrast to how stock

71

options are used at the very top of many listed companies as a force to *increase* the divide between management and shop floor.

Appropriating Ronald Coase, the previous chapter described the firm as a political response to an economic problem. This economic problem is that firms need to buy something that they cannot own and that can speak, namely labour. The solution in Coase's time was to run the firm like a dictatorship, in which employees relinquish power over their working day in exchange for a regular wage. This model first ran into problems during the 1970s, when it became clear that flexible, flatter organisational structures were more competitive than rigid, hierarchical ones. The growing significance of intangible assets makes management-as-dictatorship less and less viable. British firms have been struggling with the transition out of top-down hierarchy ever since, with varying degrees of success.

The democratic, employee-owned firm takes the principle of participation and dialogue and instates it as a constitution. The shift from industrial hierarchies to post-industrial collaborative structures cannot be achieved through rhetoric and HRM alone; it needs companies that are designed along different lines. Public services should take note as well. If the firm is a political response to an economic problem, the way in which *power* is legitimated and held to account within firms becomes a matter of economic and political urgency.

Opportunities for growth

Co-owned businesses (where employees have a stake anywhere above 25 per cent) are currently worth an estimated £25 billion to the UK economy.¹⁰⁵ Some of the larger ones, such as John Lewis Partnership and Scott Bader, were converted to employee ownership thanks to the equity being given away by the family owners several decades ago. Others were either founded or converted more recently, taking advantage of tax breaks on employee benefit trusts that have since been terminated (discussed in chapter 5). This is in addition to the extensive use of employee share ownership schemes, which a fifth of firms are now using.

Business succession is a major factor in the sector's growth. All three of the firms in the case studies presented in the next chapter are either concerned about or grappling with business succession, and see employee ownership as a solution. A 2001 survey indicates that around 30 per cent of business closures in the UK can be seen as succession failures. Selling a business is one solution, but often only produces efficiencies where the new owner strips its assets and shrinks it. Value is removed, not added.

How much room for further growth is there, and in which sectors might we expect it? Viewing the international picture, mutuality in general tends to be dominated by a handful of sectors.¹⁰⁶ Banking and insurance lend themselves to mutual models, where policy-holders, savers or borrowers have membership rights. Many of the world's largest cooperatives are in the agricultural sector. As far as employee ownership is concerned, retail and hospitality are areas where it is a significant force, unsurprisingly since customer care is cited as one of the main benefits of this ownership structure. Italy is home to a third of Europe's majority-employee-owned businesses (established as worker co-ops), and retail represents a large proportion of these.

One area with largely untapped potential for greater employee ownership is professional services and knowledgeintensive industries. These are types of business where the cooperation, commitment and innovation of employees are most important. The firm is little other than its 'human capital', and there are strong social and economic reasons why employees might expect to share in the profits of their skills. Many professional services firms, such as lawyers, accountants and architects, are currently established as partnerships, where a small minority of senior professionals have ownership and control of the firm, selling their stakes on when they leave.

Employee ownership potentially offers a different settlement to such firms. Already there are examples in professional services, such as Martin Currie investment management in Edinburgh and Make architects in London. Rewards at the very top may not be quite as great as those available to partners or founders of rival firms. Nobody retires with several million pounds. But staff turnover is lower and knowledge-sharing is higher. Where knowledge is the stock in trade, social structures must be designed to recognise this fact.

The practice of luring skilled workers with the promise of stock options became something of a joke in the wake of the dotcom boom, once it became clear that most of the options were worthless. But there is still scope for equity to play a far more significant role in employment relationships and contracts. Will Hutton has suggested that companies struggling in the recession should offer their staff shares in exchange for a pay-cut.¹⁰⁷ We saw evidence in chapter 1 that equity-based incentives had the effect of vastly increasing the amount of inequality within FTSE 100 companies. It is worth trying to imagine what it would mean for such remuneration schemes to be used with the opposite effect.

It seems that employee ownership can be especially worthwhile in those areas where the level of a firm's performance is for whatever reason unavoidably in the hands of employees. It is this that makes it a possible component of future public service reforms, especially for provision of care. Already, the maior political parties are eager to involve social enterprises in the delivery of public services, while Foundation Trust hospitals are now running effectively as independent mutuals, owned by their local communities. Examples of employee-owned public service outlets are rare but generally impressive. Greenwich Leisure is owned and run by its staff, operating seven formerly publicly-owned leisure centres, with an annual turnover of £45 million. Sunderland Home Care has a turnover of £2 million, and is now being replicated in other cities. Most strikingly, annual staff turnover at Sunderland Home Care is below 5 per cent, compared with a national average of 20 per cent among care workers.¹⁰⁸ And Central Surrey Health is owned by its 780 staff, consisting mainly of nurses and therapists.

The central argument in favour of employee ownership of public service providers is the same as it is for commercial entities. A virtuous circle of representation and productivity can be achieved, to win greater commitment and responsibility from both management and employees. As Charles Leadbeater has argued:

Staff in co-owned public service organisations frequently say they are willing and able to 'go the extra mile' to deliver a better service for people. The public sector will need more of that spirit. That is why promoting greater co-ownership should be a strategic priority for the next phase of public service renewal.¹⁰⁹

With this in mind, Jo Ellins and Chris Ham have outlined a number of ways in which employees can be given a greater stake in NHS bodies, from simply drawing them closer into decision making, through to reforming Foundation Trust hospitals with a degree of employee ownership built in.¹¹⁰

The concept of co-ownership may prove to be the most significant in future. Ownership of an organisational entity can be shared between various stakeholders and governance structures can be developed to reflect this. Founders or familyowners may wish to retain a large stake, consumers can be given a voice and dividend as in a consumer cooperative, and some equity can be sold to private shareholders. As long as the employee share is big enough to ensure representation and a significant incentive, it need not be as much as 50 per cent. Moreover, the leading examples of the co-owned model, such as Arup and Mott MacDonald, indicate that the model can scale globally very effectively. New hybrid models are constantly being developed, and this experimental approach to the ownership and governance of firms may prove to be the most fruitful for generating greater pluralism, engagement and productivity in the years ahead. It requires imaginative and well-informed accountants and lawyers, and patience while members and managers are familiarising themselves with new structures despite initial misgivings, five years on, Foundation Trusts are deemed a great success.

4 Inside the employeeowned firm

When those in positions of control seek to give power away, there is invariably a risk attached. The transition out of hierarchy can be disconcerting for all involved, and there is no certainty as to where things will end up. There is compelling economic evidence that the productivity effects of financial and democratic participation in the firm are very positive, but this does not mean it is simple to establish. This challenge is no bad thing. If, as Richard Reeves and Philip Collins argue, 'the good society is messy and unpredictable because it vests power in people', then this must be somehow reflected in the firms that such a society supports.¹¹¹

Many of the firms that adopt alternative ownership and governance structures feel that they are doing something unusual, and sometimes receive stark warnings from their professional advisers. Public understanding of employee ownership tends to rely heavily on leading examples, primarily the John Lewis Partnership, the largest employee-owned business in the UK and one of the 20 largest cooperatives in the world of any form.¹¹² Other examples, such as St Luke's advertising agency and Loch Fyne Oysters, have been celebrated in specially dedicated books.¹¹³ It is exciting to imagine what might happen, should government and expert bodies set about publicising and supporting alternative models for the firm. For the moment, however, the sector is being expanded thanks to the imagination and far-sightedness of founders and managers of businesses.

To understand a bit more about how this takes place, we will now look inside three of these firms, drawing on interviews with senior management and focus groups with staff. Each of these has either been founded as employee-owned in the last decade, or has recently converted to employee ownership. The first two are cases of indirect ownership, while the third is of direct ownership. Certain commonalities will arise, which will be drawn out and discussed later.

The creative professionals: Make architects

Lord Norman Foster is the most famous name in British architecture, having been responsible for many of the most striking construction projects of recent years, including London's GLA building, the Gherkin and Millennium Bridge. He founded Foster + Partners in the late 1960s, retaining majority ownership and control of the firm until 2007, when he sold a large minority stake in the business to 3i, the private equity fund. Behind Foster's name, however, is a very large practice of some 1,000 employees, including the architects who design these buildings in his name. Until 2003, one of the most senior and respected of these was Ken Shuttleworth.

Shuttleworth left Foster + Partners in 2003 in order to set up his own practice. But being already 51, the question of how the firm would out-live his management seemed unavoidable. Architects' practices are typically founded as partnerships by somewhat younger architects, and are then held together by the unification of ownership and control. As the partners near retirement, the question arises of who is to take over both the ownership and the control of the practice. There is no perfect answer here.

Appointing new partners from within requires the selected individuals to take on very high levels of debt to buy out their predecessors. The replacement of one elite group of owners by another can also have unfortunate consequences for workplace politics. In-fighting and suspicion over who is to succeed the partners can undermine cohesion and productivity. Having spent years or decades working in a practice owned by a small coterie of senior managers, architects find that the biggest rewards at the end are of a winner-take-all nature. This can have a perverse effect on the psychology of employees who are in direct competition for these positions for many years. Their alternative option is to leave and found their own practice. The difficulty of achieving succession from one set of partners to a new set means that it is often tempting to put the business on the market. Floatation, private equity buyouts (as with Foster + Partners) and trade sales are all effective means for the partners to cash in their stakes and solve the problem of selecting the next generation of partners. But they come with few guarantees about how the business will be managed, invested in and preserved in future.

It was with all this in mind that Shuttleworth sat down with advisers in 2003 and took out a blank piece of paper. As Barry Cooke, who was to become his financial director, remembers it:

Ken drew a little sketch of a concrete bunker without a keyhole but a key sealed inside and said what I want to do is to take ownership of this business out of my hands and put it in this concrete bunker to be absolutely bomb proof. It cannot be got at. It cannot be manipulated by future generations or by myself.

He was also aware that anyone he wanted to attract to the business would be taking a risk by leaving a job elsewhere, which needed recognising in some way. This was in addition to his concerns regarding the politics of partnership and succession in the years ahead.

Shuttleworth and Cooke were aware of employee ownership as a model – particularly having worked on projects with Arup – and asked their advisers to explore the options in this area. In 2004, Make architects was founded as a limited liability company, with a share issued to each of two trustees of an employee benefit trust. Shuttleworth and Cooke appointed themselves trustees initially, and have informally identified who their successors will be for these positions. The trust sits alongside an executive board of four directors.

Seeing as every employee, without exception, is also an owner, they are referred to internally as 'partners', and welcomed as such when they join. Make grew extremely quickly, comprising 140 employees (or partners) by the end of 2007, and is currently rated as one of the highest performing firms in a severely deflated industry.¹¹⁴ Profits have been high, and while a great deal of these are currently ploughed back into the business, surprisingly large profit share bonuses have been distributed. According to the informal constitution designed by Shuttleworth, the majority of an employee's annual bonus is set as a proportion to their salary. The remaining portion is determined by a secret ballot, in which all employees are able to allocate six votes, recognising the contribution that colleagues have made over the year. These are added up, and additional bonus payments made in proportion to votes received. This applies to every employee without exception.

In terms of governance, the firm does not use any formal mechanisms for employee representation. The trust exists to ensure that management act in the interests of employees, which includes the necessity of making profits. Since the same individuals sit on both the trust and the directors' board, there is no clear governance separation here between the management and ownership bodies.

The ownership structure produces very high levels of transparency and flat management structures. A weekly all staff meeting involves discussion of accounts and the opportunity to debate the performance and direction of the firm. In the absence of senior partners, for whom the architects would otherwise be working, hierarchies are weak and more responsibility for projects is devolved to the architect concerned.

Many partnerships are characterised by a split, whereby the partner-owner takes responsibility for the commercial side of the firm and all its projects, while the designers are paid for their drawing skills as highly skilled labourers. A wedge is driven between the financial and the creative side of the business, with the result being that the designers can often be accused of lacking business sense, while the partners can be accused of living – handsomely – off the creativity of others.

The devolved, team-based management structure of Make involves more architects taking responsibility for winning business and handling clients. This has notable productivity benefits. Fewer administrative staff are required to integrate different elements of the firm's activity, as a broader sense of responsibility is at work and projects are run holistically; clients get to deal directly with the actual designer on a project, knowing that they also have an economic stake in the company's success. Architects typically work long hours under pressure. And where some practices have to deal with staff turnover of as much as 30 per cent annually, Make's turnover is remarkably low. This makes investment in training and other intangible goods, such as giving employees paid leave to do volunteering work, substantially less risky from a business point of view.

The recession has hit architecture worse than any other profession in the UK.¹¹⁵ With projects drying up, Make made two rounds of redundancies in late 2008 and early 2009. Employee ownership clearly cannot defy a recession. The only benefits that the structure can bring in such circumstances were the following. Higher levels of transparency, with accounts shared internally, meant that economic risks could be perceived and discussed from an early stage. The economic situation is discussed openly and frankly, where every member of a workplace has equal ownership rights. Friday meetings were used to invite anyone considering a sabbatical to step forward. Individuals who were made redundant left with proper compensation as partners of a practice, and not simply employees at the whim of management.

The family business: Parfetts cash and carry

In the late 1970s, Steve Parfett was on the graduate recruitment scheme at Waitrose. His father had a background in running cash and carry businesses, and he suggested to Steve that they start a new firm together. Parfetts cash and carry was founded in Stockport in 1980, employing ten people, and turning over £1 million and a small profit in its first year. Over the next 28 years, it would grow to six depots around the North West of England, employing 650 staff and turning over £259 million annually. The industry typically works with high turnover, low profit margins and tight cost control. The company's gross profit margin is currently around 6.5 per cent.

Parfett Sr had retired in 1989, leaving Steve and his brother to run the business. By 2006, Steve himself was considering what would happen after he retired in 2012. The family's younger generation did not seem especially interested in taking over the firm. The obvious solutions were a private equity buy-out, management buy-out or – most plausibly – trade sale to a competitor. The family was not particularly keen on any of these, so Steve began to consider the model he had come across at Waitrose, part of the John Lewis Partnership.

Advisers were suspicious of the model, because they knew little about it other than the John Lewis example. But after taking advice from John Lewis and the Employee Ownership Association, and paying an auditor to assess options in this area, a solution was found. An employee benefit trust would be established, which would purchase the company with a bank loan secured against the firm's assets (mainly property). An independent evaluation was carried out, valuing the firm at £48 million with £40 million of assets, and the family agreed that they would sell to the trust for the price of the assets alone.

The sale is taking place in two stages. A £20 million loan was taken out in 2008, to purchase 55 per cent of the company on behalf of employees, which will take 15 years to pay back. The other 45 per cent of the company equity will remain with the Parfett family, until a later date when it will either be purchased with retained earnings or with more debt.

Having taken legal advice on the issue, the family decided that it would be easier to sell the company to the employees without consulting them first. But this then meant that they had the difficult task of explaining what was taking place to the employees at the six depots. A communications strategy was launched, in which senior management went round and explained the model to all staff in groups of 15–20. Initial understanding of the model was poor, though there was immediate relief amongst staff that the news was not of a sell-off.

It was clear to the Parfett family that employee ownership had governance implications, and was not simply about equity and dividends. As Steve Parfett puts it:

It's absolutely essential to shift power as well as equity. Everyone told me, you really have to mean it. As a team we're all committed to it. It's

not just a legal or financial exercise, we need to get everyone to buy into it.

Steve and his co-directors faced the challenge of designing a governance structure that acknowledged the rights of employee-owners, while also exploiting the potential for performance improvements. The structure is loosely modelled on the John Lewis example. Each depot elects a Branch Council of up to seven members, which meets twice a year with the Parfetts directors. Two members of each Branch Council also sit on a Central Council, which meets biannually and elects a representative to the board of trustees. The board of trustees sits in parallel to the board of directors, although two directors sit on both.

This is all set out in a constitutional document that outlines the corporate purpose, and which is distributed amongst all staff. As this document puts it, the Parfetts philosophy is

to continue to grow a successful and profitable business, incorporating the values and ethics of an employee owned company, and to encourage a collective responsibility, that recognises the importance of the welfare and development of both employees and customers.

Engagement with the structure varies tremendously across the workforce. Some have made an effort to get to grips with structure and stand for election. As one check-out staff-member (who reported that he had not read the constitution, as it was not for him) put it, 'I wasn't very interested in it at first, but the longer I'm here the more I think I might get involved. Put myself down for the council or something.' Many start to take a greater long-term interest in the direction and performance of the company. Others view the new governance arrangements as obscure, and continue to fear the disruption that they expect Steve Parfett's retirement will cause.

Most staff feel that the new structure is not working as well as it needs to, and feel that it is a challenge for them and for management to sort this out. The anxiety about the transition is having positive spillovers, as consultation and transparency is increasing to deal with it. Inevitably, there is a limit to how much democracy is desirable before it creates a sense of a 'talking shop' – some staff report that the councils are too big and meet for too long. Low-level grievances exist, for instance that the elections are a popularity contest and that the Central Council is too full of 'ties' (as in the garment). But this is all part of the difficult process of empowering employees, many of whom are in comparatively low skilled work.

This is a curious transition in some respects. As one employee put it, 'The model was attractive to us because it didn't represent an enormous change from how the business had been run under family ownership.' The employees felt that they would continue to be valued, just as the family felt that the business they'd built would survive. One of the most tangible aspects of employee ownership will not be operating for a number of years: dividend payments will only begin once the debt has been paid off. Employees already get a Christmas bonus of an additional week's pay, which persists under employee ownership. Other than the bedding in of the new governance structures, not a great deal has changed so far.

But once the last family members retire, the debt is paid off and the democratic practices are embedded, the culture and the politics of the firm will surely be very different. The director who stands to succeed Steve Parfett as managing director is aware that he will have much higher accountability to his employees than the founders of the business ever had.

The human capital collective: Quintessa Consultants David Hodgkinson is a scientist who has worked at the interface of science and nuclear policy since the 1970s. In 1987, he was invited to establish the environmental consultancy division of a company called Intera Information Technologies, to specialise in advising governments on nuclear energy. He did so, and it grew rapidly, to his great satisfaction. He saw that there was real value in

creating something from nothing. One day you have an idea, a year later you're paying people, they're paying their mortgages, you have something tangible. By 1997, his division had offices around the world, employing 70 expert staff. But after twelve years developing the business, it was sold to Enviros. He decided to start all over again, but in such a way that the same outcome could not arise.

In 1999, Hodgkinson re-mortgaged his house to finance the founding of Quintessa, initially retaining 100 per cent of the equity. He was adamant that the company was not going to admit equity finance from outsiders who did not respect the long-term integrity of the firm. At the centre of his vision of the firm was the expertise and talent of those he intended to hire, and the ownership and governance structure would stem from this. Furthermore, anyone who did join his new business would necessarily be giving up secure employment elsewhere and taking some risk, especially when Quintessa was in its early stages.

With this in mind, he decided to offer equity stakes in the business to employees, of which there are now 22 in the UK and 10 in Japan. As he explains:

Employee ownership was chosen primarily for cultural reasons. Certainly within a company of talented individuals, it's only right that it is employee-owned.

Those who joined him immediately were given larger shares than those who arrived later, with Hodgkinson now holding just over half of the equity. The shares pay a dividend, and in addition an annual bonus is paid to all staff, but neither of these is very large as a proportion of total remuneration. Shares cannot be sold, except back to the firm on departure, whereupon they are valued by outside accountants.

The company now has an annual turnover of £2.4 million, and its sales and profit growth are higher than the largest competitor in its field.¹¹⁶ Additional data are unavailable, but management report no recorded absenteeism and only one member of staff has ever left during the firm's ten-year history.

A company of this size and nature has clear opportunities to be run in a cooperative fashion – there are company directors, but no managers as such. Having out-sourced most of their administrative functions, nearly all of the staff are scientists and mathematicians, who might otherwise be working in academic jobs. They share a common vocation to employ their scientific expertise in a responsible, commercially competitive fashion. As one director explains, 'it's a quasi-academic, quasi-commercial environment.' An informal ethos of collaboration permeates the firm, with a strong recognition that there is more to work, to business and to science than making money.

The most important function of employee ownership is that it represents an obstacle to excessive interference or exploitation by outside financiers. Hodgkinson has received around ten offers to sell the company, which he has declined. Many of the directors and staff have worked for companies with external shareholders, and witnessed the pressures that they can exert over a company's direction. The business is profitable, but not governed by the pursuit of profit. As one employee puts it,

We can afford to take a longer-term view, which is relevant to how we work as individuals. We can go through a quiet patch, or invest in ourselves, but as long as the company is sustainable, we can do that.

Given their dedication to highly specialist fields of knowledge, the intangible assets that are being invested in and drawn on cannot have a price put on them, so the company is not for sale. This is the factor that Quintessa staff are most adamant about. The dividend payment on shares is more significant symbolically than economically, and none of the staff feels that it is necessary as an incentive to work hard or well. Meanwhile, the firm has little need of outside capital, given that its directors have little desire to expand the business much further, and it is accumulating cash, which may contribute to buying Hodgkinson's shares from him at a later date.

Quintessa will face a business succession issue when Hodgkinson retires in 2014 or thereabouts. His shares will have to be bought from him at some stage. Quintessa directors are considering shifting ownership towards a trust-based model (indirect ownership) though they worry about the tax disadvantages, as discussed in the next chapter. This is something that remains to be worked out.

A different political economy

These case studies were chosen for their diversity, so as to give an indication of the flexibility of ownership options and plurality of benefits. Parfetts is in a relatively low-skilled sector, while the other two examples employ highly skilled professionals. Make distributes large sums of profit, which has a major impact on the psychology of its employees, while the other two do not attach great significance to the dividend. Quintessa has explicit ethical, even ideological reasons for valuing this ownership model, whereas the other two are more focused on long-term profitability. Parfetts has taken great pains to establish a new, representative governance structure, based on a formal constitution, whereas the other two firms have allowed greater participation to permeate their culture in less formal ways. But it is possible to extract some commonalities.

Firstly, these case studies give an insight into the psychology of ownership, which plays out in what others might term 'financial literacy'. Chapter 2 cited the standard definition, as given by the OECD, that ownership is made up of the right to control and the right to a dividend. It was argued that ownership is more complex than this, especially where there are intangible assets at stake, which cannot be *entirely* controlled or privatised. The owners in these three cases are given not only a set of rights, but also certain responsibilities for the firm. The notion of transparency suggests that they have new freedoms to peer inside the firm's workings, which is true; but they also have a great deal more financial information thrust upon them, whether they like it or not.

In some respects, being subservient in the workplace is a comfortable if frustrating condition. The opaqueness and dishonesty of management is a good excuse for not participating further. The firms described in this chapter remove this excuse, by disseminating information about their accounts and plans for the future. In the case of Parfetts, where staff are not professional experts, they were largely unfamiliar with the technicalities of corporate governance, business planning and accounting. But all staff are being encouraged or dragged into considering these issues, because as owners they have a responsibility to know more about them. In the case of Make, financial and creative responsibilities are handled simultaneously by the same individuals, meaning that architects become literate in financial and business issues at a much earlier stage of their careers. They share directly in the business rewards for their creativity, but they also take greater responsibility for winning business and satisfying commercially-minded clients.

Secondly, these examples give a close-up view of how democratic and economic accounts of autonomy can reinforce one another. As the previous chapter outlined, it is not implausible for the two to come into conflict. Excessive worker control *can* lead to productivity problems.¹¹⁷ Pushing too much responsibility for business performance onto employees *can* lead to stress and unhappiness.

In the case of these businesses, employees recognise the benefits of being in a relatively secure environment, with less hierarchy than would be the case under different ownership models. They are less at the whim of shareholders or the managers who strive to satisfy them. In the case of Parfetts, this had long been the case, with the family viewed as benign patrons, but employee ownership represented an equally benign alternative to takeover by a rival firm. From this position of relative security, satisfaction and voice, greater effort and innovation comes more naturally, with less resentment or stress. Contrary to what is often assumed, risk-taking and innovation can occur within stable secure environments, just not where this is achieved through strong hierarchy.

Finally, we can see in these examples that the appeal of this model cannot be easily categorised as either economic or ethical. It is a different means of organising power within the firm, with different consequences for the distribution of risk and reward. It can properly be described as a different political economy. In knowledge-intensive sectors, it has particularly striking implications, reducing the 'winner-take-all' elements of many professional services firms, which effectively enable a minority to profit hugely off the knowledge of the majority.

Employee ownership can certainly deliver higher productivity and job satisfaction. What it is unlikely to deliver is a vast windfall of wealth for a small minority of individuals. The Parfett family will extract £40 million from their firm, but only after building it over 30 years, and even then this is less than the market valuation of £48 million. Ken Shuttleworth is by far the greatest beneficiary of Make's ownership and governance model (remember that dividends are proportionate to salary), but he will not be able to sell a controlling share to a private equity company as Norman Foster did. David Hodgkinson will eventually extract his shares in Quintessa for a large sum of money, but for half as much as he might have done.

None of these individuals is a philanthropist, such as the ones that gave away ownership of John Lewis Partnership or Scott Bader. They are each very wealthy and their businesses are profitable. The trade-off that is made is between viewing a firm as an item of property to be traded, versus viewing it as a complex social entity consisting of intangible assets to be organised, invested in, profited from and enjoyed. Where the latter is prioritised, ways must be found of locking value into the firm, just as better ways *should* have been found to lock value into former building societies.

5 Enabling a new model of the firm

The financial crisis and subsequent recession made establishing a new regulatory framework into a priority, particularly in the financial services sector. A new political economy seems imminent, built on a reworked relationship between the state and the market and on new sources of wealth creation within the market. The government has important choices to make in determining which organisational forms and economic practices are to grow in the wake of the current crisis. As the evolutionary economist Carlota Perez argues:

It is not enough to define the criteria for good regulation, it is also crucial to define the criteria for a whole set of other policies that must accompany the treatment of finance in order to guide market behaviour in the direction that is most beneficial for all (including an effective and profitable financial world).¹¹⁸

Unsurprisingly given the urgency of the financial situation, little thought has yet been paid to the types of institutional structures that will sit at the heart of the next wave of capitalism in Britain or elsewhere. The regulatory imperatives, focused firstly on restoring the functionality of banking and secondly on reducing systemic risk, have taken rightful priority over the broader, longer-term questions of the socio-economic ecology of the future.

As chapter 1 stressed, times of crisis open up opportunities to factor in broader cultural, political and technological priorities into economic designs. Crises are not only moments for the conflicts within capitalism to be dealt with, as Karl Marx argued, but also opportunities to address conflicts between a variant of capitalism and the civil society in which it sits. While the principle of the firm's accountability to shareholders will and must survive, there is scope to temper this with the addition of accountability to other stakeholders, and also to nurture other models of the firm altogether.

Employee ownership and mutuals are only one part of any vision of Britain's economic future, and will not single-handedly resolve the four crises outlined in chapter 1. On the other hand, alternative models of ownership and control can make a positive contribution to addressing the four crises discussed in chapter 1.

#1: The banking crisis

Britain needs to pursue diversity in the way that its financial services are controlled and delivered. While the pursuit of very high returns, achieved through *proportionately* high levels of risk-taking, has its own legitimate place within the overall ecology of the financial services industry, greater pluralism is needed. The government has a one-off opportunity to deliver some of this, as it currently owns large shares in former building societies such as Northern Rock.

A return to mutualism is now necessary, with asset locks properly constructed to avoid another wave of demutualisation. Simpler, lower-risk banking is not possible where banks are focused on their share price, which leads to the loss of accountability. For those individuals who feel most alienated from gargantuan and unaccountable financial institutions, the future may involve a return to even more traditional forms of credit and debt, such as credit unions, Zopa's peer-to-peer model, or collective equity investments facilitated by the internet.

#2. The competitiveness crisis

We have seen convincing evidence that employee ownership leads to improved business performance. It is a structure that appears well suited to innovative and knowledge-intensive businesses, where information-sharing and employee commitment is key. Where indirect models of ownership are concerned, they defend long-term investment in human and social capital from the short-term demands of external shareholders. In which case, the obvious question is – why aren't there more of them around? Part of the explanation lies in the lack of suitable financing options. Venture capital and other forms of private equity are scarcely any more sympathetic to the long-term challenges of value creation than institutional investors. Establishing employee ownership is heavily dependent on debt finance and specialist equity funds such as Baxi Partnership.

But the overwhelming reason why this model isn't more prevalent is that it is inadequately understood by managers, policy makers and legal and accountancy professionals. Employee ownership tends to be associated either with distribution of shares and options (which is one way of extending ownership, but a comparatively weak and exclusive one) or it is associated with worker-managed co-ops, which summons up visions of 1970s inefficiency. The full range of mutual and coowned structures is not properly appreciated, and their productivity benefits are over-looked. This needs to change.

#3. The fiscal crisis

Those who work for employee-owned firms recognise that they are more transparent, civic-minded and committed to high levels of service.¹¹⁹ They generate positive externalities, in the form of happier, healthier employees and surrounding communities. This is why firms' internal dynamics and governance structures matter politically. Sustaining progressive goals and social cohesion through an era of lower state spending requires careful thought about the social effects of decentralised economic institutions. Why is it that talk of community and devolution of power always gravitates towards the third sector and social enterprise? There are choices to be made about how profit-making, economic institutions are to be structured as well.

In addition to this, the scope to employ employee-owned and hybrid models in the delivery of public services needs to be explored further, and the productivity benefits evaluated. There are various areas of public service, such as Royal Mail, Network Rail and Forensic Science, where hybrid models of ownership and control – drawing multiple stakeholders into a single structure – offer solutions.

#4. The moral crisis

Chapter 1 outlined a range of evidence, demonstrating the contribution that the shareholder value paradigm has made to Britain's growing inequality. Inequality is not simply a product of amorphous 'market forces' or 'globalisation', but a symptom of the choices we make about the organisation of risk, reward and power within our dominant economic structures. In their landmark analysis of inequality, *The Spirit Level*, Richard Wilkinson and Kate Pickett conclude that democracy and greater equality *within* the workplace is one of the best hopes for limiting or reversing the broader societal trends of the last 30 years. The authors have since established the Equality Trust, which proposes that

greater economic democracy is essential in order to transform our economy, reduce income differences by bringing pay differentials under democratic control, redistribute wealth and create the foundations for a healthier, happier and more sustainable society.¹²⁰

This is an area where political engagement is most important. We have overwhelming evidence of the malign effects of inequality. We also know that many of these effects, especially regarding health and mental health, fall to the government to act upon.

There is no clear established evidence regarding the pay differentials under alternative models of ownership and corporate governance, but two things are clear. The use of stock options, concentrated in a small minority of hands, does nothing for business performance, but does a great deal to exacerbate already high levels of inequality. Secondly, employee representation, which leads to accountability of management to staff, *at the very least* opens up an avenue for debate about remuneration practices within the firm.

An agenda for the firm

With these points in mind, there are various things that can be done to inject much-needed diversity into Britain's conventional models of ownership and control.

Altering tax incentives

Tax incentives are by far the most powerful of the government's tools for influencing company ownership and financial structure. As we saw in chapter 3, there are already a number of widely used employee share schemes that are encouraged through tax advantages. Tweaking the tax system not only creates narrowly economic incentives to distribute ownership, but also performs a 'signalling' role in alerting and educating businesses, accountants and lawyers about new schemes.

The most significant impediment to the growth of employee-owned businesses, at least where indirect ownership is concerned, is the 2003 reform which removed tax advantages from the creation of employee benefit trusts (EBTs), save where shares were then transferred to employees. This was done because the existing scheme was being abused as a form of tax avoidance. But as the managers of the case-studied firms stressed repeatedly, this makes the indirect ownership model significantly less attractive. The Employee Ownership Association has put forward a proposal in which the existing (tax advantaged) Share Incentive Plan (SIP) could be expanded to include EBTs.¹²¹ Given the clear social and economic advantages of indirect ownership forms over direct forms, as discussed in chapter 3, any politician or policy maker seeking to support growth in this area should treat this as the highest priority.

The cost of restoring the tax advantages to EBTs is estimated by the Employee Ownership Association at £51 million. Obviously the Treasury will want to know where else this money could be found. Current tax advantaged share schemes cost the exchequer an estimated £1.2 billion and If the government were to back the EBT model, the money could be saved by reducing the advantages of the SAYE or EMI schemes.¹²² The SAYE does not incentivise long-term shareholding, because it is most widely used in listed companies where employees can dispose of their shares as soon as they exercise their option. The EMI scheme, aimed at small businesses and start-ups, is now so successful that it is unusually costly to the Treasury, averaging £28,000 per employee. Money could be diverted from either of these into EBT tax advantages, while EMI's continuation could be made conditional on the creation of EBTs.

Restoring and defending mutualism in banking

The government should heed the proposals being made by the new Centre for Mutual and Employee-Owned Business at Oxford University, to consider re-mutualising Northern Rock. HM Treasury/UFI has various options for divesting its current stake in the bank, but only a return to a mutualised building society model will offer any greater diversity and stability to the financial sector as a whole. Mutualisation must be defended with secure asset locks, to prevent future demutualisations after the present banking crisis has passed.

Facilitating finance for mutual and employee-owned business

Businesses that are only indirectly owned (ie do not have shareholders) face difficulties getting access to finance. They are either dependent on debt finance from banks, or on niche areas of private equity financing such as Baxi Partnership. At present, banks do not properly understand the employee-owned model, and too many venture capitalists associate it with social enterprise, and therefore with non-profit-making ventures. Aggressive venture capital practices are, in any case, unsuited to this sector, as one of its main virtues is that these businesses are run in pursuit of *long-term* business performance, drawing on expert *insider* management knowledge.

The government should intervene financially to support growth in the sector. It could do so via Regional Development Agencies, enterprise capital funds, the Post Office and its influence over the banks in which it has major stakes. There is no reason to believe that a dedicated public venture capital fund for mutuals and employee-owned businesses, run along similar principles to the Baxi Partnership, could not yield a return to the taxpayer. The market failure here is in equity finance for organisations that are profit making but not profit maximising, and operate according to longer financial time horizons than orthodox venture capital and private equity can currently support.

Piloting mutualised public service providers

Innovation in the public sector must include innovations in organisational forms. Independently managed public interest companies – such as Foundation Trust hospitals and Channel 4 – are already recognised public service providers. There is a compelling case for the NHS to contract with employee-owned and co-owned businesses in the delivery of care services. Hybrid organisations, co-owned by government, employees and users, are viable models that need piloting in areas of health, education and transport.

Spreading the benefits of share ownership

As chapter 3 explained, Britain has had a succession of schemes to encourage share ownership amongst employees, dating back to the Thatcher government. However, equity-based remuneration remains something of an elitist benefit. It is difficult to imagine it ever becoming a force for greater equality, because the low-paid cannot afford to divert nearly as much (if any) of their income into shares as the highly paid. Moreover, those who rely on week-to-week wages should not have their remuneration diverted into year-to-year equity schemes.

That said, the government should look into ways of extending participation in such schemes, as a way of spreading asset ownership. Extending schemes to *all staff*, with a guarantee that wages will not be cut, could be a statutory requirement for firms seeking to benefit from the tax advantages of SIPs, SAYE or CSOP. The government should also consider the Employee Share Ownership Plan (ESOP) model in the USA, in which employee savings and pensions are put into the firm in order to grow a *collective* employee stake in the business. The ESOP model is vulnerable to the charge that it puts all employee 'eggs in one basket', and should be treated with some caution, as the bankruptcy of ESOP-owned United Airlines suggests. The French example dating back to Charles de Gaulle, where profitsharing is mandatory for firms above 50 employees, is another means of increasing cooperation between capital and labour.

At present, external investors in small businesses and startups are being subsidised by the taxpayer, thanks to the EMI. Meanwhile, capital gains tax relief for internal investors in startups only applies to those owning more than 5 per cent of the equity. By spreading these forms of tax relief to include *employees* willing to invest small amounts of risk capital in new ventures, the government can encourage a different approach to start-ups, in which the risks and rewards are more evenly distributed. Risk-taking and entrepreneurship can exist within the firm, and not only via more arms-length relationships. The cases of Quintessa and Make give a sense of the impact this can have. The cost of this could also be covered by restricting the reach of the EMI scheme.

The agenda for professional services

Many of the obstacles which restrict the growth of employee ownership are due to inadequate expertise in this area. Lawyers and accountants work around the assumption that such businesses are inefficient or not-for-profit. The Centre for Mutual and Employee-Owned Business intends to educate future business leaders on the reality and merits of alternative models. But professions and professional services firms have an important contribution to make. The potential benefits of this model within professional services mean that, for instance, an employee-owned law firm would be able to demonstrate and communicate the viability and productivity potential of a flatter structure. One of the best signals that alternative models are to be taken seriously would be for a major professional services firm to convert from partnership to an employee benefit trust model.

Celebrating new models

Politicians and policy makers can support the sector by giving it the recognition and clarity of analysis that it deserves. There are still a number of basic misunderstandings in this area. Most commonly, as cited above, the profit-making, indirect ownership model is overlooked in favour of employee share ownership on the one hand (typically recognised as a form of collective profitsharing) and the cooperative movement or social enterprise on the other (typically recognised as less business-like). Equally, the 'eggs in one basket' argument, that employee ownership is excessively risky, is applied inappropriately, often drawing on American examples of ESOPs.

Government can signal support for this sector via the tax scheme and by privileging mutual and employee-owned models when it contracts with the private sector, as described above. But politicians could also offer more vocal support and clarity regarding the various options. One of the first things that a sympathetic government could do in this area would be to commission in-depth, comparative quantitative research on the relationship between ownership models and various externalities, such as the health, well-being and civic engagement of employees. Qualitative evidence would suggest that listed companies do worse in this area than alternative ownership models, because of their focus on share value. If this is backed up by quantitative evidence, this would suggest that Britain's overall political economy is not nearly as efficient, *in the aggregate*, as the gurus of shareholder value would have us believe.

Establishing and converting more firms to these models

At present, founders, managers and owners of firms who wish to establish employee ownership are reliant on their own imagination and a few sources of information. The case studies in chapter 4 suggest that the model is still perceived as somewhat unusual. However, entrepreneurs and businesses facing succession issues are strongly encouraged to consider alternatives to floatation, private equity buyouts and trade-sales, for all the reasons contained in this report. Anyone minded to do this is encouraged to contact either the Employee Ownership Association (www.employeeownership.co.uk) or Baxi Partnership (www.baxipartnership.co.uk).

Managers and owners confronting a problem of business succession or considering a transition to employee ownership should refer to publications *Employee Ownership as a Solution to Business Succession* and *Shared Company: How employee ownership works*, both of which are available on the Employee Ownership Association website.

Notes

- Committee on Oversight and Government Reform, 'Testimony of Dr Alan Greenspan', 23 Oct 2008, http://oversight.house.gov/ documents/20081023100438.pdf (accessed 26 May 2009).
- 2 'Welch condemns share focus price', *Financial Times*, 12 Mar 2009.
- 3 R Bruner, *Deals from Hell: M&A lessons that rise above the ashes* (Hoboken, NJ: Wiley 2005).
- 4 JK Galbraith, *The New Industrial State* (Princeton, NJ: Princeton University Press 2007).
- 5 This evolutionary approach to capitalism is most famously advanced in the work of Joseph Schumpeter – see J Schumpeter, *Capitalism, Socialism and Democracy* (London and New York: Routledge, 1976).
- 6 See N Hsieh, 'Justice in production', *Journal of Political Philosophy* 16, no 1 (2008); and Hsieh, 'Freedom in the workplace' in S White and D Leighton (eds), *Building a Citizen Society: The emerging politics of republican democracy* (London: Lawrence & Wishart, 2008).
- 7 G Mulgan, 'After capitalism', Prospect, Apr 2009.
- 8 See G Arrighi, *The Long Twentieth Century* (London: Verso, 2009); C Freeman and F Louca, *As Time Goes By: From the Industrial Revolution to the Information Revolution* (Oxford: Oxford University Press, 2001).

- 9 The term 'publicly owned' means that shares in these companies are traded among members of the public via stock markets. It does not mean owned by the state. It stands in contrast to 'privately owned' (where ownership is retained within the firm's management, founders or buyers).
- 10 See L Mitchell, *The Speculation Economy: How finance triumphed over industry* (San Francisco: Berrett-Koehler, 2007).
- 11 C Freeman and F Louca, *As Time Goes By*.
- 12 See L Boltanski and E Chiapello, *The New Spirit of Capitalism* (London: Verso, 2006).
- 13 M Wolf, 'Reform of regulation has to start by altering incentives', *Financial Times*, 23 Jun 2009; see also L Bebchuk and H Spamann, 'Regulating bankers' pay', Harvard Law and Economics Discussion Paper 641 (2009), http://ssrn.com/ abstract=1410072.
- 14 TUC, TUC Fund Manager Voting Survey 2009, 2009.
- 15 C Dillow, 'Why aren't hedge funds failing as fast as banks?', *The Times*, 17 Sep 2008.
- 16 See D Llewellyn, *Re-mutualisation: A systematic strategy after the crisis* (Oxford: Kellogg College, 2009, forthcoming).
- 17 See J Lanchester, 'It's finished', *London Review of Books*, 28 May 2009.
- 18 This is stressed in R Ayadi et al, Investigating Diversity in the European Banking Industry: Performance and role of savings banks (Brussels: Centre for European Policy Studies, 2009).
- 19 Editorial, *Financial Times*, 27 Apr 1999.
- 20 Office of National Statistics, *Blue Book*, 2008.

- 21 OECD figures cited in keynote address by Már Gudmundsson, deputy head of the Monetary and Economic Department of the BIS, at the Financial Technology Congress 2008, Boston, 23 Sep 2008.
- 22 'Personal debt in UK exceeds GDP for second year', *Independent*, 22 Aug 2008.
- 23 P Mandelson, RSA lecture, 17 Dec 2008.
- 24 'Warning over renewables as economic crisis leaves funding gap', *Guardian*, 21 Mar 2009.
- 25 BERR, *The 2008 Productivity and Competitiveness Indicators*, www.berr.gov.uk/files/file49953.pdf.
- 26 'Mandelson calls for more state activism to replace "light touch", *Independent*, 5 Mar 2009.
- 27 See 'Darling to blame boards for bank meltdown', *Guardian*, 16 Jun 2009; and Paul Myners' speech to British Bankers Association Annual Conference, 30 Jun 2009, www.hmtreasury.gov.uk/speech_fsst_300609.htm.
- 28 For an outstanding sociological analysis of this tension see L Boltanski and L Thevenot, On Justification: Economies of worth (Princeton, NJ: Princeton University Press 1991).
- 29 P Hall and D Soskice, *Varieties of Capitalism: The institutional foundations of comparative advantage* (Oxford: Oxford University Press, 2001).
- 30 R Reeves, 'Progressive austerity: an agenda to protect the poor', *Financial Times*, 21 May 2009.
- 31 S White and D Leighton (eds), Building a Citizen Society.

- 32 OECD, *Growing Unequal: Income distribution and poverty in OECD countries* (Paris: Organisation for Economic Co-operation and Development, 2008).
- 33 S Hampson and R Reeves, *Paths to Equality* (London: Demos, forthcoming).
- 34 M Brewer et al, *Poverty and Inequality in the UK: 2009* (London: Institute for Fiscal Studies, 2009).
- **35** R Barwell et al, 'The distribution of assets, income and liabilities from UK households: results from the 2005 NMG Research survey', *Bank of England Quarterly Bulletin*, spring 2006.
- **36** Office of National Statistics, *Social Trends 2006* (London: Office of National Statistics, 2006).
- 37 See R Wilkinson and K Pickett, *The Spirit Level: Why more equal* societies almost always do better (London: Penguin, 2009).
- 38 The World Database of Happiness, http://worlddatabaseofhappiness.eur.nl.
- **39** M Brewer et al, *Racing Away? Income inequality and the evolution of high incomes* (London: Institute for Fiscal Studies, 2008).
- 40 See P Hall and D Soskice, *Varieties of Capitalism*; and A Pendleton, 'The liberal market model of finance, ownership and governance: an evaluation of its effect on labour', *Asia Pacific Journal of Human Resources* 47, no 2 (2009).
- 41 'Pay gap widens between board and shop floor', *Guardian*, 30 Aug 2007.
- 42 Ibid.
- 43 R Dore, Stock Market Capitalism, Welfare Capitalism: Japan and Germany versus the Anglo-Saxons (Oxford: Oxford University Press, 2000).

- 44 See for example J Pierce et al, 'The state of psychological ownership: integrating and extending a century of research', *Review of General Psychology* 7, no 1 (2003); this is also the basis of John Locke's theory of property.
- 45 The work of legal theorist Tony Honore, identifying ten different rights and duties that are associated with ownership, has been frequently cited here. See T Honore, 'Ownership' in *Making Law Bind: Essays legal and philosophical* (Oxford: Oxford University Press, 1987).
- 46 OECD, *Principles of Corporate Governance* (Paris: Organisation for Economic Co-operation and Development, 2004).
- 47 Cited in Tomorrow's Company, *Tomorrow's Owners: Stewardship of Tomorrow's Company* (London: Tomorrow's Company, 2008).
- 48 In the 'coordinated market economies' such as Germany and Japan, it is not unusual for banks to be represented on the boards of companies to which they lend, however.
- 49 C Leadbeater, A Piece of the Action: Employee ownership, equity pay and the rise of the knowledge economy (London: Demos, 1997).
- 50 R Coase, 'The nature of the firm', *Economica* 4 (1937).
- 51 J Rifkin, The Age of Access (London: Penguin, 2000).
- 52 J Donzelot, 'The mobilisation of society' in G Burchell et al, *The Foucault Effect: Studies in governmentality* (New York: Simon & Schuster, 1991).
- 53 Tomorrow's Company, Tomorrow's Owners.
- 54 S Davis et al, *The New Capitalists* (Boston, MA: Harvard Business School Press, 2006).
- 55 J Kay, 'Test of possession', *Financial Times*, 28 Feb 1997.

- 56 Tomorrow's Company, Tomorrow's Owners.
- 57 S Davis et al, The New Capitalists.
- 58 See the excellent blog http://labourandcapital.blogspot.com/.
- 59 Tomorrow's Company, Tomorrow's Owners.
- 60 British Venture Capital Association figures available at www.bvca.co.uk/About-BVCA/features/FAQs (accessed 2 Sep 2009).
- 61 T Jenkinson, 'Private equity' in *The European Economic Advisory Group Report on the European Economy* (Munich: Ifo Institute for Economic Research, 2009).
- 62 Tomorrow's Company, Tomorrow's Owners.
- 63 BERR, The 2008 Productivity and Competitiveness Indicators.
- 64 Cited in Tomorrow's Company, Tomorrow's Owners.
- 65 See Co-operative Action, Co-operative Capital: A new approach to investment in co-operatives (Manchester: Co-operative Action, 2004).
- 66 Mutuo, Mutuals Yearbook 2008 (Herts: Mutuo, 2008).
- 67 Table from Mutuo, Mutuals Yearbook 2008.
- 68 See R Bruner, Deals from Hell.
- 69 M Wilson and C Mills, *Co-operative Values Make a Difference in the Curriculum and the Governance of Schools* (Manchester: The Co-operative College, 2007).
- 70 P Miller and P Skidmore, Disorganisation: Why future organisations must 'loosen up' (London: Demos, 2004).

- 71 J Michie and M Sheehan, 'Labour "flexibility" securing management's right to manage badly?' in B Burchell et al (eds), Systems of Production: Markets, organisations and performance (London: Routledge, 2003).
- 72 BERR, The 2008 Productivity and Competitiveness Indicators.
- 73 N Hsieh, 'Justice in production'.
- 74 R Sennett, The Corrosion of Character: The personal consequences of work in the new capitalism (New York and London: Norton, 1998); and R Sennett, The Craftsman (London: Allen Lane, 2008).
- 75 World Health Organisation, *Closing the Gap in a Generation: Health equity through action on the social determinants of health* (Geneva: World Health Organisation, 2008).
- 76 D Coats and R Lehki, 'Good Work': Job quality in a changing environment (London: The Work Foundation, 2008).
- 77 J Goddard, 'A critical assessment of the high-performance paradigm', *British Journal of Industrial Relations* 42, no 2 (2004).
- 78 A Felstead et al, *Skills at Work 1986 to 2006* (Oxford: ESRC Centre for Skills, Knowledge and Organisational Performance, 2007).
- 79 C Shirky, *Here Comes Everybody: The power of organising without organisations* (London: Allen Lane, 2008).
- **80** P Bradwell and R Reeves, *Network Citizens: Power and responsibility at work* (London: Demos, 2008).
- 81 See R Erbmann et al, *Collective Power: Changing the way we consume energy* (London: Co-operative Party, 2009).
- 82 See S White, "Revolutionary liberalism"?: the philosophy and politics of ownership in the post-war Liberal Party', *British Politics* 4, no 2 (2009).

- 83 E Kaarsemaker et al, *Employee Share Ownership Plans: A review* (York: York Management School, University of York, 2009).
- 84 The relative distinction between 'employee ownership' and 'coownership' is borrowed from the Employee Ownership Association. There are clearly overlaps between all these definitions.
- 85 See for example Mutuo, 21st Century Pioneers: The co-operative group constitutional review (London: Mutuo, 2007).
- 86 A Bryson and R Freeman, 'How does shared capitalism affect economic performance in the UK?' (Cambridge, MA: National Bureau of Economic Research, 2008).
- 87 'Companies with tax-advantaged employee share schemes', HMRC (2008)
- 88 J Gates, The Ownership Solution (London: Penguin, 1998).
- 89 See S White, *The Civic Minimum* (Oxford: Oxford University Press, 2003); and S White and D Leighton, *Building a Citizen Society*.
- 90 S White, "Revolutionary Liberalism"?'
- 91 D Kruse, 'Research evidence on the prevalence and effects of employee ownership', *Journal of Employee Ownership Law and Finance* 14, no 4 (2002).
- 92 P Burns, Good Business: The employee ownership experience (London: JOL, 2006)
- 93 A Pendelton, 'Employee participation in employee share ownership: an evaluation of the factors associated with participation and contributions in Save As You Earn plans', *British Journal of Management* (2009 forthcoming).

- 94 P Burns, Good Business.
- 95 Co-operatives UK and JOL, 'Employee ownership as a solution to business succession' (London: Co-operatives UK and JOL, 2005).
- 96 'Employee owned companies shares perform better in the long term according to Employee Ownership Index', press release, Field Fisher Waterhouse, 22 Dec 2008
- 97 Oxera, Tax Advantaged Employee Share Schemes: Analysis of productivity effects. Report 2: Productivity measured using gross value added (Oxford: Oxera, 2007).
- 98 See A Bryson and R Freeman, 'How does shared capitalism affect economic performance in the UK?'; D Kruse, 'Research evidence on the prevalence and effects of employee ownership'; and C Doucouliagos, 'Worker participation and productivity in labor-managed and participatory capitalist firms: a metaanalysis', *Industrial and Labor Relations Review* 49 (1995).
- **99** A Bryson and R Freeman, 'How does shared capitalism affect economic performance in the UK?'
- 100 P Burns, Good Business; All Party Parliamentary Group on Employee Ownership, Share Value: How employee ownership is changing the face of business (London: All Party Parliamentary Group on Employee Ownership, 2008).
- 101 See J Michie and C Oughton, 'HRM, employee share ownership and corporate performance', *Research and Practice in Human Resource Management* 11, no 1 (2003).
- 102 R Freeman et al, 'Monitoring colleagues at work: profit sharing, employee ownership, broad-based stock options and workplace performance in the United States', CEP discussion paper 647 (London: Centre for Economic Performance, London School of Economics, 2004).

- 103 See J Michie and M Sheehan, 'No innovation without representation? An analysis of participation, representation, R&D and innovation', *Economic Analysis: Journal of Enterprise and Participation* 2, no 2 (1999).
- 104 All Party Parliamentary Group on Employee Ownership, *Share Value*; and P Burns, *Good Business*.
- 105 This estimate is made by the Employee Ownership Association, who suspect that it is much too low.
- 106 See International Co-operative Alliance, Global 300 List 2008: The world's major co-operatives and mutual businesses (Geneva: International Co-operative Alliance, 2008).
- 107 W Hutton, 'When the boss asks you to take a pay cut, demand a share in the company', *Observer*, 21 Jun 2009.
- 108 J Ellins and C Ham, NHS Mutual: Engaging staff and aligning incentives to achieve higher levels of performance (London: The Nuffield Trust, 2009).
- 109 C Leadbeater, *Innovation Included: Why co-owned businesses are* good for public services (London: Employee Ownership Association, 2007).
- 110 J Ellins and C Ham, *NHS Mutual*.
- 111 R Reeves and P Collins, *The Liberal Republic* (London: Demos, 2009).
- 112 International Co-operative Alliance, *Global 300 List 2008*.
- 113 A Law, Open Minds: 21st century business lessons from St Lukes (New York and London: Texere, 2001); D Erdal, Local Heroes: How Loch Fyne oysters embraced employee ownership and business success (New York: Viking, 2008).

- 114 Plimsoll, The Plimsoll Analysis: A comprehensive review of the top 1000 companies in the UK architects industry (Stockton-on-Tees: Plimsoll Publishing, 2009).
- 'Architect job losses soar as crunch hits construction', *Guardian*, 20 Mar 2009
- 116 Plimsoll, The Plimsoll Analysis: Environmental consultants analyses the UK environmental consultants market and the 669 largest companies in the industry (Stockton-on-Tees: Plimsoll Publishing, 2009).
- 117 This is confirmed in C Doucouliagos, 'Worker participation and productivity in labor-managed and participatory capitalist firms', which otherwise makes a strong economic case for workforce empowerment.
- 118 C Perez, 'After crisis: creative construction', *Open Democracy*, 5 Mar 2009.
- 119 P Burns, Good Business.
- 120 Taken from the Equality Trust's website, www.equalitytrust.org.uk/take-action/economic-democracy (accessed 28 Jul 2009).
- 121 All Party Parliamentary Group on Employee Ownership, *Share Value*.
- 122 Employee Ownership Association, internal estimates.

References

'Architect job losses soar as crunch hits construction', *Guardian*, 20 Mar 2009.

'Darling to blame boards for bank meltdown', *Guardian*, 16 Jun 2009.

'Mandelson calls for more state activism to replace "light touch", *Independent*, 5 Mar 2009.

'Pay gap widens between board and shop floor', *Guardian*, 30 Aug 2007.

'Personal debt in UK exceeds GDP for second year', *Independent*, 22 Aug 2008.

'Warning over renewables as economic crisis leaves funding gap', *Guardian*, 21 Mar 2009.

'Welch condemns share focus price', *Financial Times*, 12 Mar 2009.

All Party Parliamentary Group on Employee Ownership, *Share Value: How employee ownership is changing the face of business* (London: All Party Parliamentary Group on Employee Ownership, 2008).

Arrighi, G, The Long Twentieth Century (London: Verso, 2009).

Ayadi, R, et al, *Investigating Diversity in the European Banking Industry: The performance and role of savings banks* (Brussels: Centre for European Policy Studies, 2009). Barwell, R, et al, 'The distribution of assets, income and liabilities from UK households: results from the 2005 NMG Research survey', *Bank of England Quarterly Bulletin*, spring 2006.

Bebchuk, L, and Spamann, H, 'Regulating bankers' pay', Harvard Law and Economics Discussion Paper 641 (2009), at http://ssrn.com/abstract=1410072.

BERR, *The 2008 Productivity and Competitiveness Indicators*, at www.berr.gov.uk/files/file49953.pdf.

Boltanski, L, and Chiapello, E, *The New Spirit of Capitalism* (London: Verso, 2006).

Boltanski L, and Thevenot, L, *On Justification: Economies of worth* (Princeton, NJ: Princeton University Press 1991).

Bradwell, P, and Reeves, R, *Network Citizens: Power and responsibility at work* (London: Demos, 2008).

Brewer, M, et al, *Poverty and Inequality in the UK: 2009* (London: Institute for Fiscal Studies, 2009).

Brewer, M, et al, *Racing Away? Income inequality and the evolution of high incomes* (London: Institute for Fiscal Studies, 2008).

Bruner, R, *Deals from Hell: M&A lessons that rise above the ashes* (Hoboken, NJ: Wiley 2005).

Bryson, A, and Freeman, R, 'How does shared capitalism affect economic performance in the UK?' (Cambridge, MA: National Bureau of Economic Research, 2008).

Burns, P, Good Business: The employee ownership experience (London: JOL, 2006).

Coase, R, 'The nature of the firm', *Economica* 4 (1937).

Coats, D, and Lehki, R, 'Good Work': Job quality in a changing environment (London: The Work Foundation, 2008).

Committee on Oversight and Government Reform, 'Testimony of Dr Alan Greenspan', 23 Oct 2008, at http://oversight.house.gov/documents/20081023100438.pdf (accessed 26 May 2009).

Co-operative Action, *Co-operative Capital: A new approach to investment in co-operatives* (Manchester: Co-operative Action, 2004).

Co-operatives UK and JOL, 'Employee ownership as a solution to business succession' (London: Co-operatives UK and JOL, 2005).

Davis, S, et al, *The New Capitalists* (Boston, MA: Harvard Business School Press, 2006).

Dillow, C, 'Why aren't hedge funds failing as fast as banks?', *The Times*, 17 Sep 2008.

Donzelot, J, 'The mobilisation of society' in G Burchell et al, *The Foucault Effect: Studies in governmentality* (New York: Simon & Schuster, 1991).

Dore, R, Stock Market Capitalism, Welfare Capitalism: Japan and Germany versus the Anglo-Saxons (Oxford: Oxford University Press, 2000).

Doucouliagos, C, 'Worker participation and productivity in labor-managed and participatory capitalist firms: a metaanalysis', *Industrial and Labor Relations Review*, 49 (1995).

Ellins, J, and Ham, C, *NHS Mutual: Engaging staff and aligning incentives to achieve higher levels of performance* (London: The Nuffield Trust, 2009).

Erbmann, R, et al, *Collective Power: Changing the way we consume energy* (London: Co-operative Party, 2009).

Erdal, D, Local Heroes: How Loch Fyne oysters embraced employee ownership and business successs (New York: Viking, 2008).

Felstead, A, et al, *Skills at Work 1986 to 2006* (Oxford: ESRC Centre for Skills, Knowledge and Organisational Performance, 2007).

Freeman, C, and Louca, F, *As Time Goes By: From the Industrial Revolution to the Information Revolution* (Oxford: Oxford University Press, 2001).

Freeman, R, et al, 'Monitoring colleagues at work: profit sharing, employee ownership, broad-based stock options and workplace performance in the United States', CEP Discussion Paper 647 (London: Centre for Economic Performance, London School of Economics, 2004).

Galbraith, JK, *The New Industrial State* (Princeton, NJ: Princeton University Press 2007).

Gates, J, The Ownership Solution (London: Penguin, 1998).

Goddard, J, 'A critical assessment of the high-performance paradigm', *British Journal of Industrial Relations* 42, no 2 (2004).

Hall, P, and Soskice, D, *Varieties of Capitalism: The institutional foundations of comparative advantage* (Oxford: Oxford University Press, 2001).

Hampson, S, and Reeves, R, *Paths to Equality* (London: Demos, forthcoming).

Honore, T, *Making Law Bind: Essays legal and philosophical* (Oxford: Oxford University Press, 1987).

Hutton, W, 'When the boss asks you to take a pay cut, demand a share in the company', *Observer*, 21 Jun 2009.

Hsieh, 'Freedom in the workplace' in S White and D Leighton (eds), *Building a Citizen Society: The emerging politics of Republican democracy* (London: Lawrence & Wishart, 2008).

Hsieh, N, 'Justice in production', *Journal of Political Philosophy* 16, no 1 (2008).

International Co-operative Alliance, *Global 300 List 2008: The world's major co-operatives and mutual businesses* (Geneva: International Co-operative Alliance, 2008).

Jenkinson, T, 'Private equity' in *The European Economic Advisory Group Report on the European Economy* (Munich: Ifo Institute for Economic Research, 2009).

Kaarsemaker, E, et al, *Employee Share Ownership Plans: A review* (York: York Management School, University of York, 2009).

Kruse, D, 'Research evidence on the prevalence and effects of employee ownership', *Journal of Employee Ownership Law and Finance* 14, no 4 (2002).

Lanchester, J, 'It's finished', *London Review of Books*, 28 May 2009.

Law, A, *Open Minds: 21st century business lessons from St Lukes* (New York and London: Texere, 2001).

Leadbeater, C, A Piece of the Action: Employee ownership, equity pay and the rise of the knowledge economy (London: Demos, 1997).

Leadbeater, C, Innovation Included: Why co-owned businesses are good for public services (London: Employee Ownership Association, 2007). Llewellyn, D, *Re-mutualisation: A systematic strategy after the crisis* (Oxford: Kellogg College, 2009, forthcoming).

Michie, J, and Oughton, C, 'HRM, employee share ownership and corporate performance', *Research and Practice in Human Resource Management* 11, no 1 (2003).

Michie, J, and Sheehan, M, 'Labour "flexibility" – securing management's right to manage badly?' in B Burchell et al (eds), *Systems of Production: Markets, organisations and performance* (London: Routledge, 2003).

Michie, J, and Sheehan, M, 'No innovation without representation? An analysis of participation, representation, R&D and innovation', *Economic Analysis: Journal of Enterprise and Participation* 2, no 2 (1999).

Miller, P, and Skidmore, P, *Disorganisation: Why future* organisations must 'loosen up' (London: Demos, 2004).

Mitchell, L, *The Speculation Economy: How finance triumphed over industry* (San Francisco: Berrett-Koehler, 2007).

Mulgan, G, 'After capitalism', Prospect, Apr 2009.

Mutuo, 21st Century Pioneers: The co-operative group constitutional review (London: Mutuo, 2007).

Mutuo, Mutuals Yearbook 2008 (Herts: Mutuo, 2008).

Myners, Paul, speech to British Bankers Association Annual Conference, 30 Jun 2009 at www.hmtreasury.gov.uk/speech_fsst_300609.htm.

OECD, Growing Unequal: Income distribution and poverty in OECD countries (Paris: Organisation for Economic Co-operation and Development, 2008).

OECD, *Principles of Corporate Governance* (Paris: Organisation for Economic Co-operation and Development, 2004).

Office of National Statistics, *Social Trends 2006* (London: Office of National Statistics, 2006).

Oxera, Tax Advantaged Employee Share Schemes: Analysis of productivity effects. Report 2: Productivity measured using gross value added (Oxford: Oxera, 2007).

Pendelton, A, 'Employee participation in employee share ownership: an evaluation of the factors associated with participation and contributions in Save As You Earn plans', *British Journal of Management* (2009 forthcoming).

Pendleton, A, 'The liberal market model of finance, ownership and governance: an evaluation of its effect on labour', *Asia Pacific Journal of Human Resources* 47, no 2 (2009).

Perez, C, 'After crisis: creative construction', *Open Democracy*, 5 Mar 2009.

Pierce, J, et al, 'The state of psychological ownership: integrating and extending a century of research', *Review of General Psychology* 7, no 1 (2003).

Plimsoll, *The Plimsoll Analysis: A comprehensive review of the top 1000 companies in the UK architects industry* (Stockton-on-Tees: Plimsoll Publishing, 2009).

Plimsoll, The Plimsoll Analysis: Environmental consultants analyses the UK environmental consultants market and the 669 largest companies in the industry (Stockton-on-Tees: Plimsoll Publishing, 2009).

Rappaport, A, Creating Shareholder Value: The new standard for business performance (New York: Free Press, 1986).

Reeves, R, 'Progressive austerity: an agenda to protect the poor', *Financial Times*, 21 May 2009.

Reeves, R, and Collins, P, *The Liberal Republic* (London: Demos, 2009).

Rifkin, J, The Age of Access (London: Penguin, 2000).

Schumpeter, J, *Capitalism, Socialism and Democracy* (London and New York: Routledge, 1976).

Sennett, R, The Corrosion of Character: The personal consequences of work in the new capitalism (New York and London: Norton, 1998).

Sennett, R, The Craftsman (London: Allen Lane, 2008).

Shirky, C, *Here Comes Everybody: The power of organising without organisations* (London: Allen Lane, 2008).

Tomorrow's Company, *Tomorrow's Owners: Stewardship of Tomorrow's Company* (London: Tomorrow's Company, 2008).

TUC, TUC Fund Manager Voting Survey 2009, 2009.

White, S, "Revolutionary Liberalism"?: the philosophy and politics of ownership in the post-war Liberal Party', *British Politics* 4, no 2 (2009).

White, S, *The Civic Minimum* (Oxford: Oxford University Press, 2003).

White, S, and Leighton, D (eds), *Building a Citizen Society: The emerging politics of Republican democracy* (London: Lawrence & Wishart, 2008).

Wilkinson, R, and Pickett, K, *The Spirit Level: Why more equal societies almost always do better* (London: Penguin, 2009).

Wilson, M, and Mills, C, *Co-operative Values Make a Difference in the Curriculum and the Governance of Schools* (Manchester: The Co-operative College, 2007).

Wolf, M, 'Reform of regulation has to start by altering incentives', *Financial Times*, 23 Jun 2009.

World Health Organisation, *Closing the Gap in a Generation: Health equity through action on the social determinants of health* (Geneva: World Health Organisation, 2008).

Demos - Licence to Publish

The work (as defined below) is provided under the terms of this licence ('licence'). The work is protected by copyright and/or other applicable law. Any use of the work other than as authorized under this licence is prohibited. By exercising any rights to the work provided here, you accept and agree to be bound by the terms of this licence. Demos grants you the rights contained here in consideration of your acceptance of such terms and conditions.

1 Definitions

- A 'Collective Work' means a work, such as a periodical issue, anthology or encyclopedia, in which the Work in its entirety in unmodified form, along with a number of other contributions, constituting separate and independent works in themselves, are assembled into a collective whole. A work that constitutes a Collective Work will not be considered a Derivative Work (as defined below) for the purposes of this Licence.
- B 'Derivative Work' means a work based upon the Work or upon the Work and other preexisting works, such as a musical arrangement, dramatization, fictionalization, motion picture version, sound recording, art reproduction, abridgment, condensation, or any other form in which the Work may be recast, transformed, or adapted, except that a work that constitutes a Collective Work or a translation from English into another language will not be considered a Derivative Work for the purpose of this Licence.
- c 'Licensor' means the individual or entity that offers the Work under the terms of this Licence.
- D 'Original Author' means the individual or entity who created the Work.
- **E** 'Work' means the copyrightable work of authorship offered under the terms of this Licence.
- F 'You' means an individual or entity exercising rights under this Licence who has not previously violated the terms of this Licence with respect to the Work,or who has received express permission from Demos to exercise rights under this Licence despite a previous violation.

2 Fair Use Rights

Nothing in this licence is intended to reduce, limit, or restrict any rights arising from fair use, first sale or other limitations on the exclusive rights of the copyright owner under copyright law or other applicable laws.

3 Licence Grant

Subject to the terms and conditions of this Licence, Licensor hereby grants You a worldwide, royalty-free, non-exclusive,perpetual (for the duration of the applicable copyright) licence to exercise the rights in the Work as stated below:

- to reproduce the Work, to incorporate the Work into one or more Collective Works, and to reproduce the Work as incorporated in the Collective Works;
- B to distribute copies or phonorecords of, display publicly,perform publicly, and perform publicly by means of a digital audio transmission the Work including as incorporated in Collective Works; The above rights may be exercised in all media and formats whether now known or hereafter devised. The above rights include the right to make such modifications as are technically necessary to exercise the rights in other media and formats. All rights not expressly granted by Licensor are hereby reserved.

4 Restrictions

The licence granted in Section 3 above is expressly made subject to and limited by the following restrictions:

- A You may distribute, publicly display, publicly perform, or publicly digitally perform the Work only under the terms of this Licence, and You must include a copy of, or the Uniform Resource Identifier for, this Licence with every copy or phonorecord of the Work You distribute, publicly display, publicly perform, or publicly digitally performYou may not offer or impose any terms on the Work that alter or restrict the terms of this Licence or the recipients' exercise of the rights granted hereunder.You may not sublicence the WorkYou must keep intact all notices that refer to this Licence and to the disclaimer of waranties.You may not distribute, publicly display, publicly perform, or publicly digitally perform the Work with any technological measures that control access or use of the Work in a manner inconsistent with the terms of this Licence Agreement.The above applies to the Work apart from the Work itself to be made subject to the terms of this Licence. If You create a Collective Work, upon notice from any Licencor You must, to the extent practicable, remove from the Collective Work any reference to such Licensor or the Original Author, as requested.
- B You may not exercise any of the rights granted to You in Section 3 above in any manner that is primarily intended for or directed toward commercial advantage or private monetary

compensation. The exchange of the Work for other copyrighted works by means of digital filesharing or otherwise shall not be considered to be intended for or directed toward commercial advantage or private monetary compensation, provided there is no payment of any monetary compensation in connection with the exchange of copyrighted works.

c If you distribute, publicly display, publicly perform, or publicly digitally perform the Work or any Collective Works, You must keep intact all copyright notices for the Work and give the Original Author credit reasonable to the medium or means You are utilizing by conveying the name (or pseudonym if applicable) of the Original Author if supplied; the title of the Work if supplied. Such credit may be implemented in any reasonable manner; provided, however, that in the case of a Collective Work, at a minimum such credit will appear where any other comparable authorship credit.

5 Representations, Warranties and Disclaimer

- A By offering the Work for public release under this Licence, Licensor represents and warrants that, to the best of Licensor's knowledge after reasonable inquiry.
 - Licensor has secured all rights in the Work necessary to grant the licence rights hereunder and to permit the lawful exercise of the rights granted hereunder without You having any obligation to pay any royalities, compulsory licence fees, residuals or any other payments;
 - ii The Work does not infringe the copyright, trademark, publicity rights, common law rights or any other right of any third party or constitute defamation, invasion of privacy or other tortious injury to any third party.
- B except as expressly stated in this licence or otherwise agreed in writing or required by applicable law,the work is licenced on an 'as is'basis,without warranties of any kind, either express or implied including,without limitation,any warranties regarding the contents or accuracy of the work.

6 Limitation on Liability

Except to the extent required by applicable law, and except for damages arising from liability to a third party resulting from breach of the warranties in section 5, in no event will licensor be liable to you on any legal theory for any special, incidental consequential, punitive or exemplary damages arising out of this licence or the use of the work, even if licensor has been advised of the possibility of such damages.

7 Termination

- A This Licence and the rights granted hereunder will terminate automatically upon any breach by You of the terms of this Licence. Individuals or entities who have received Collective Works from You under this Licence,however, will not have their licences terminated provided such individuals or entities remain in full compliance with those licences. Sections 1, 2, 5, 6, 7, and 8 will survive any termination of this Licence.
- B Subject to the above terms and conditions, the licence granted here is perpetual (for the duration of the applicable copyright in the Work). Notwithstanding the above, Licensor reserves the right to release the Work under different licence terms or to stop distributing the Work at any time; provided, however that any such election will not serve to withdraw this Licence (or any other licence that has been, or is required to be, granted under the terms of this Licence), and this Licence will continue in full force and effect unless terminated as stated above.

8 Miscellaneous

- Each time You distribute or publicly digitally perform the Work or a Collective Work, Demos
 offers to the recipient a licence to the Work on the same terms and conditions as the licence
 granted to You under this Licence.
- B I^F any provision of this Licence is invalid or unenforceable under applicable law, it shall not affect the validity or enforceability of the remainder of the terms of this Licence, and without further action by the parties to this agreement, such provision shall be reformed to the minimum extent necessary to make such provision valid and enforceable.
- c No term or provision of this Licence shall be deemed waived and no breach consented to unless such waiver or consent shall be in writing and signed by the party to be charged with such waiver or consent.
- D This Licence constitutes the entire agreement between the parties with respect to the Work licensed here. There are no understandings, agreements or representations with respect to the Work not specified here. Licensor shall not be bound by any additional provisions that may appear in any communication from You. This Licence may not be modified without the mutual written agreement of Demos and You.

This project was supported by:

John Lewis Partnership

The financial crisis has called into question many of our core assumptions about economic structures, governance and institutions. But there has been little attention paid to the basic unit of economic collaboration and production: the firm. In recent decades Britain developed a corporate monoculture in which the 'shareholder value' creed treated firms simply as the property of their shareholders, to be traded, exploited and disposed of in pursuit of profit.

Government policy making has done little to call this culture into question, depriving our economy of a richer vision of what a good company is and what it can do. This crisis is a chance to ask deep questions about our firms: how can they meet social and political as well as economic goals? How can firms be modelled so that not only shareholders but employees, the economy and society profit?

Many of these models already exist. Mutual and employee-owned models of business operate with longer time-horizons, achieving higher levels of performance and customer satisfaction. They nurture greater power for individuals over their economic lives and increase the accountability of managers. This report argues it is time to bring these models out of the wilderness and into the debate about where capitalism goes next. Presenting a wide range of quantitative data alongside three new case studies of employee-owned firms, it offers a new vision of economic autonomy where democratic companies drive a happier and more sustainable economy.

William Davies is a Demos Associate and a Research Fellow at the Institute for Science, Innovation & Society, Said Business School.

ISBN 978-1-906693-22-0 **£10** © Demos **2009**

